review of income and wealth

Review of Income and Wealth Series 65, Number S1, November 2019 DOI: 10.1111/roiw.12436

HOUSEHOLD FOOD INSECURITY AND MATERNAL AND CHILD NUTRITIONAL STATUS: EVIDENCE FROM MAHARASHTRA

by Chandana Maitra

School of Economics, The University of Sydney, Sydney, Australia

VANI SETHI

United Nations Children Fund India Country Office, New Delhi, India

SAYEED UNISA

Department of Mathematical Demography and Statistics, International Institute for Population Sciences, Mumbai, India

AND

SRIRAM SHANKAR

Centre for Social Research and Methods & Research School of Economics, The Australian National University, Canberra, Australia

In this paper we rigorously examine the association of household food security status with maternal and child nutritional status exploring a rich micro-level database, the Comprehensive Nutrition Survey Maharashtra 2012. Using Household Food Insecurity Access Scale (HFIAS) as a metric to measure household food security, we apply simultaneous probit models to estimate the effect of food insecurity on maternal and child nutritional outcomes. The modelling framework addresses possible endogeneity in the above relationship. Findings indicate that household food insecurity increases the risk of maternal and child underweight with no effect on child stunting, wasting or maternal overweight. Women's decision-making power mediates these associations. Food security interventions should be effective in tackling the concerns with undernutrition, however, food alone is not the solution. A mix of nutrition-specific and nutrition-sensitive policies is warranted. Food security interventions should be integrated into policies for human development.

JEL Codes: I15, C31, D10, I30, O15

Keywords: experiential, endogeneity, food access, human development, nutritional status, women's bargaining power

Note: We are grateful to UNICEF and IIPS for access to the rich microlevel data from Comprehensive Nutrition Survey Maharashtra, 2012. We sincerely acknowledge the input received from Dr. Mark Nord (The Voices of the Hungry Project, FAO) in guiding us with the statistical evaluation of the food security indicator. We also thank our three anonymous referees and the participants of IARIW-ICRIER Conference on "Experiences and Challenges in Measuring Income, Inequality and Poverty in South Asia" (held at New Delhi, India, on November 2017) for their comments and suggestions which greatly enriched the paper. The usual disclaimer applies.

Correspondence to: Chandana Maitra, School of Economics, Level 5, Social Sciences Building [A02], The University of Sydney, NSW 2006, Sydney (e:chandana.maitra@sydney.edu.au)

1. INTRODUCTION

The nature and extent of the association of food insecurity with nutritional status is of interest both from the perspective of economic theory and policy making. Research in the developed countries (primarily US) has evidence that lack of access to sufficient quantity of safe and nutritious food is associated with children's health, behavioral, academic and emotional problems beginning as early as infancy (Skalicky et al., 2006; Cook and Frank, 2008; Gundersen et al., 2011). The potential channel linking food insecurity to the above outcomes could be malnutrition which encompasses both overweight and undernutrition including micronutrient deficiencies; and has direct adverse consequences in terms of disability, disease, brain development, learning and earnings (Victora et al., 2008; Black et al., 2008). However, the association of food insecurity with nutritional outcomes is not yet well understood (FAO, 2018). Nutrition is a social determinant of health and development (WHO, 2000). Rich longitudinal evidence exists on the economic benefits of avoiding undernutrition (Victora et al., 2008). Food insecurity might be a distant cause of malnutrition (Murray and Lopez, 1999). Therefore, understanding the exact nature of the relationship not only facilitates effective policy formulation but also contributes to the discourse in the development literature on the pathways to inclusive growth via poverty reduction and human development. Current research sheds some light on this question, however, further rigorous research is warranted (Maitra, 2018). The task is complex and is also getting increasingly challenging given the newly emerging complexities such as multiple burden of malnutrition (Haddad et al., 2015; WHO, 2017) in the global nutrition scenario. Furthermore, the literature investigating food insecurity-malnutrition linkage suffers from methodological concerns relating to selection bias, measurement errors and unobserved heterogeneity which often render food insecurity endogenous with respect to nutritional outcomes. Limited availability of panel data further complicates empirical research. A recent review of 120 studies on the link between experiential food insecurity and nutritional status of adults and children across the globe (Maitra, 2018) identifies the failure to address these methodological issues as a crucial gap in the literature. The few studies that examine the above relationship in the Indian context (Mukhopadhya and Biswas, 2011; Gupta et al., 2013; Chandrasekhar et al., 2017) offer rich insight on the nexus between the two phenomena, albeit with similar limitations.

Given the above, in the present paper, we rigorously examine whether household food insecurity increases the risk of malnutrition in women and children. We aim to address the current research gap by modelling the above relationship with due consideration to the concerns relating to endogeneity. We undertake this exercise using a rich household level cross sectional dataset, Comprehensive Nutrition Survey Maharashtra (CNSM) 2012 (IIPS-UNICEF, 2013). The dataset is based on a representative sample of 2650 children aged 0–23 months and their mothers. The key indicator of household food insecurity (HFI) in our study is Household Food Insecurity Access Scale (HFIAS) - an experienced-based food security scale (EBFSS) which is a direct measure of access to food (Coates *et al.*, 2007). The nutritional outcomes are captured by anthropometric indicators—stunting (too short for one's age), wasting (too thin for one's height) and underweight in children; and BMI status for women (underweight/overweight) (WHO, 2010).

^{© 2019} International Association for Research in Income and Wealth

The intellectual motivation of our query stems from the fact that if food insecurity has any causal association with nutritional outcomes it can be theoretically modelled as an input in the production of human capital, given the robust microeconomic evidence on the intergenerational consequences of malnutrition on learning outcomes (Lewis, 2009; Grantham-Mcgregor et al., 2007). A malnourished mother gives birth to a malnourished child and "severe early malnutrition, particularly during the period of rapid brain growth in the first two years of life, presents a permanent structural insult to brain function, leading to irreversible intellectual impairment" (Levitsky and Strupp, 1995, cited in Fishman et al., 2004, p. 101). Such strong linkage between nutrition and human development suggests almost an immediate and direct link between food security and human development, provided we see a close alignment of food insecurity with nutritional outcomes. Such a possibility allows us to conceive of food security within the framework of Sen's (1985) capability approach (see Burchi and Muro, 2016), since "the human development discourse is conceptually underpinned by the capability approach" (Osmani, 2016, p. 2). The interaction between entitlement, capability and human development is also relevant in this context. Food insecurity can be both a cause and consequence of entitlement failure (Sen, 1981). It may result in entitlement failure (reduced command over commodities including food) by making an individual less capable in the labor market via the channels of poor nutrition, poor health and poor cognitive development. There exists a potential feed-back effect from malnutrition to the capacity of providing food, health, education and care (Burchi and Muro, 2016). This perspective also allows us to place food security within the broad realm of well-being, all the possible activities and social expressions a person would realize-his or her 'beings and doings' (Sen, 1985). The latter is also the idea of 'agency' as conceived by Sen (1999). Within this framework, we can extend the concept of food security in a manner that it ultimately strengthens human security-ensuring security by investment in human development, not in arms (UNDP, 1994). The usefulness of such a framework is, it has long term implications for food security policy formulation. It leads us away from the sole focus on entitlement, income or livelihood and allows us to think about food insecurity as a cause and consequence of lack of education, health and other basic capabilities that constitute people's wellbeing. This approach is people-centric and it also helps us to embrace the 'stability' aspect of food security (UNDP, 1994) which has the element of time embedded in it -- "capability to be food secure", having a longer-term perspective (Burchi and Muro, 2016).

The above intellectual pursuit is further motivated by the fact that the food insecurity-malnutrition link reflects an underlying conceptual association between the *access* and *utilization* dimensions of food security. With the introduction of UNICEF's conceptual model (UNICEF, 1990) (Fig. B.1)¹ which provides a comprehensive framework to understand the causes of malnutrition, the definition of food security was broadened to include *utilization* as an additional element along with *availability* and *access* to food in the World Food Summit 1996 (WFS, 1996). Conceptually, food utilization relates to the capacity of an individual to absorb the nutrients in the food consumed and is determined by practices, beliefs, eating

¹Figure B.1 presents an adapted version of UNICEF framework reported in Black et al. (2008).

^{© 2019} International Association for Research in Income and Wealth

habits, hygiene, sanitation and health (WFS, 1996). Ideally, proper utilization of food should be reflected in good nutritional status. The UNICEF framework identifies food insecurity as a necessary but not sufficient condition for nutrition security because malnutrition may be caused by other inter-related factors-inadequate maternal and child care, insufficient health services and unsanitary environment. This conceptual evolution of food security is parallel to the move from "entitlement" to basic human "capabilities" in the conversation on hunger –from command over food to capability to be free from hunger (Drèze and Sen, 1989) which argues, "a more reasoned goal would be to make it possible to have the capability to avoid undernourishment and escape deprivations associated with hunger" (Drèze and Sen, 1989, p. 13). This shift of focus to "nutritional capabilities" reflects the recognition that access to food is not adequate and utilization is critical (Burchi and Muro, 2016).

The implication of the above discussion for our purpose is, it generated confusion on the logical difference between food and nutrition security and subsequently shifted the unit of analysis from household to individual. One aspect of this development is, it further broadens the concept of food security by recognizing the possibilities of inequalities in intra-household distribution of food. When household is the unit of analysis in measuring access to food, the intra-household dynamics remains unexplored because adequate access to food at the household level does not automatically translate to access at the individual level (Maxwell and Smith, 1992). An indirect way of measuring individual food access would then be to look at anthropometric indicators which are often considered the gold standard measures of nutritional status (Jones et al., 2013); and are outcomes of inadequate food access (Frankenberger, 1992) given the other conditions (care and health environment) remain stable. The other implication of the above development is purely policy-focused, related to targeting and monitoring. The more closely aligned are the access and utilization components of food security (adequate access-adequate food intake-better utilization-better nutritional status), stronger is our ability to detect intra-household disparities in access to food. In this scenario, policy can isolate and target those individuals who happen to be malnourished in an otherwise food secure environment. On a similar note, if changes in food security status are directly related to changes in nutritional status, then food security interventions should automatically improve nutritional outcomes. Nutritional relevance of food security indicators is also established (Headey and Ecker, 2013) ensuring that these indicators are able to identify nutritionally vulnerable individuals at minimal additional cost of data collection.

Given this policy relevance of our query it is not surprising that the volume of literature investigating the link between food security and nutrition security is growing. Our study is a contribution to this literature not just because it entails methodological improvement, but also because we take a novel perspective on the query by situating it within the framework of capability approach. This analytic framework offers a longer-term potential for analyzing food security as an underlying determinant of human development, a discourse which may have started with Sen's conversation on hunger (1981; 1985), however, barring some rare attempts (Burchi and Muro, 2016; UNDP, 2013) has not gained momentum since then.

^{© 2019} International Association for Research in Income and Wealth

Maharashtra provides us a perfect setting to investigate the above queries because of the perplexing dissociation between indicators of growth, nutrition and food security the state experienced. Current statistics reports that Maharashtra is the leading Indian state in terms of per capita income, with the State economy forecasted to grow by 7.3 percent during 2017-18 over the previous year-much above the 6.5 percent expected growth rate for India in 2017 (Economic Survey Maharashtra, 2018). However, the nutritional indicators perform dismally. Maharashtra falls into the category of very high prevalence of wasting (≥ 15 percent) and underweight (\geq 30 percent) (Menon *et al.*, 2017). Simultaneously, the nutrition scenario is characterized by prevalence of high rates of maternal anemia (48 percent in 2016) which in fact escalated during 2006–2016 (NFHS, 2017). Prevalence of maternal underweight was also very high-above the national average of 36 per cent in 2006. While the proportion of underweight women has declined in 2016, that of overweight women has increased substantially (NFHS, 2017). The combination of these trends indicates the emergence of intra-household multiple burden of malnutrition such as stunting in child and obesity in mother in the same household; or intra-individual multiple burden such as stunting, obesity and anemia in the same individual (Meenakshi, 2016). The food security scenario has been equally grim with the state experiencing one of the highest rates of calorie deficiency among the Indian states in 2004–05 (NSSO, 2007). In 2008, Maharashtra ranked ten out of 17 Indian states following the Global Hunger Index (von Grebmer et al., 2008), and was listed in the category of 'alarming hunger'. What seems to be puzzling is that since 2006 the nutritional status of children has improved due to sharp decline in the prevalence of child stunting; however, food insecurity continues to be a major concern (Haddad et al., 2014). The paradox signals an apparent lack of correspondence between the two phenomena. These anomalies provide us the final motivation to undertake this research.

In modelling the food insecurity-malnutrition relationship we used the framework of simultaneous probit models, more specifically recursive bivariate probit/ ordered probit models (Greene, 2012), to account for the endogeneity of experiential food insecurity, our key explanatory variable in the models. We also undertook a statistical exercise to test the reliability and validity of HFIAS before proceeding with any statistical analysis based on the indicator, which is also an improvement over the existing studies. Our findings indicate that HFI is a significant risk factor for child and maternal underweight, however, HFI has no effect on the risk of child stunting and maternal overweight. The impact of HFI on the risk of wasting disappears once the effect of women's decision-making power (a proxy for women's bargaining power in the household) is incorporated into the models. Women's bargaining power seems to be moderating the relationship between food insecurity and nutritional outcomes, highlighting the importance of addressing the concerns with intra-household distribution of food. Overall, the findings direct toward the need to alleviate household food access for the purpose of addressing nutritional deprivation of mothers and children residing in the household. However, long term nutritional deprivation of children (reflected in stunting) as well as the association with overweight requires further investigation, perhaps in the light of multiple burden of malnutrition. In general, a mix of nutrition-specific and nutrition-sensitive policies is suggested for tackling malnutrition.

^{© 2019} International Association for Research in Income and Wealth

The paper has been organized as follows: Section 2 discusses method which includes details of survey methodology, data, variables and statistical analysis; Section 3 describes the conceptual framework of the study, Section 4 reports results, Section 5 discusses the findings and Section 6 concludes with policy recommendations and direction for future research.

2. Method

2.1. Data, Study Design and Participants

The study is based on CNSM 2012 (IIPS-UNICEF 2013). The survey aimed at assessing the nutritional status of children aged 0-23 months through anthropometric measurements and infant and young child feeding practices in rural and urban areas of the state. The selection of sample was done separately in rural and urban areas using a multi-stage stratified sampling procedure. The rural sample was selected in two stages, with the selection of Primary Sampling Units (PSUs), villages, at the first stage. This stage was followed by a random selection of households within each PSU, in the second stage, among those households where at least one child under two years is residing. In urban areas, a threestage sampling procedure was followed; selecting wards at the first stage, Census Enumeration Blocks at the second stage, and households with at least one child under two years at the third stage. The sample size was determined in terms of the number of under-two children as they were the focus of the survey. Remaining details on survey and sampling strategies are available in CNSM report (IIPS-UNICEF, 2013). Data collection was carried out during February-April 2012. The final sample comprises 2,630 households and 2650 children.

2.2. Survey Instrument

CNSM used three types of questionnaires: Household Questionnaire, the Mother's Questionnaire and Child's Questionnaire. The Household Questionnaire interviewed either the head of the household or any adult member of the household available at the time of the survey. The section on food security was administered to the household member primarily involved in food preparation. Information was collected on age, sex, relationship to the head of the household and marital status for each person listed. Additionally, information was gathered on the main source of drinking water, type of toilet facility, source of lighting, type of cooking fuel, religion and caste/tribe of the household head, ownership of a house, ownership of agricultural land, ownership of livestock and ownership of other selected items. Mother's Questionnaire was administered to all women in the household who have at least one (living) child born after January 1, 2010. Mother's profile including her age, marital status, age at marriage, literacy status, educational attainment and work status were collected. Information were also collected on mother's exposure to media, decision making and involvement in community organizations such as Self-Help Groups and; maternal health such as pregnancy, fertility history, antenatal care received, food intake, nutritional status and; lifestyle indicators such as tobacco use and

^{© 2019} International Association for Research in Income and Wealth

alcohol consumption. For every child born after January 1, 2010, separate questionnaire was administered to the mother. The child's questionnaire collected information on child characteristics such as birth date, birth order and sex of the child. CNSM also included questions on infant and young child feeding practices including breastfeeding status, frequency of breastfeeding in 24 hours prior to survey, complementary foods given to the child in 24 hours prior to the survey and their frequency.

2.3. Variables

Dependent Variables: Anthropometric Indicators of Nutritional Status of Mothers and Children

Following WHO (2010) guidelines, three standard indices of physical growth that describe the nutritional status of children are constructed: 1) height-for-age z-scores; 2) weight-for-height z-scores; and 3) weight-for-age z-scores.² Since, in the present paper we are interested in examining the *risk* of malnutrition due to food insecurity, we express the scores as categorical variables. Children with height-for-age z-score <-2 standard devitations (SD) from the median of the reference population are considered short for their age or stunted. Stunting reflects longer term nutritional deprivation often aggravated by illness and is often intergenerational. Children with weight-for-height z-score <-2 SD from the median of the reference population are classified as thin for their height or wasted. Wasting captures short term nutritional deficiency which might be a result of inadequate nutrition and disease in the recent past. Weight-for-age is a composite index representing the sum of the information given by the other two indices, height-forage and weight-for-height. Weight-for-age can be low because of stunting (short stature) and/or wasting (recent weight loss). Children with weight-for-age z-score <-2 SD from the median of the reference population are underweight.³

The ordered categorical variables *HAZ*, *WHZ* and *WAZ* are created to denote three categories of nutritional status–healthy ($-2 \le z$ -score $\le +2$), moderate ($-3 \le z$ -score ≤ -2) and severe (z-score ≤ -3), respectively. Accordingly, we've the following representation of variables:

 $HAZ = \begin{cases} 0 & \text{if healthy} \\ 1 & \text{if modertaely stunted} \\ 2 & \text{if severely stunted} \end{cases}$

²It is important to note that anthropometric indicators are observable markers for the internal, physiological state of nutriture termed as malnutrition (Fishman *et al.*, 2004).

³Children over two standard deviations from the median weight-for-height of the WHO Child Growth standards are overweight. We do not model the association of household food insecurity with overweight status of children because proportion of overweight children in the sample is not large enough to allow a meaningful investigation. To be consitent, only children with z scores less than equal to +2 SD are included in the analysis. Children with height-for-age z-scores below -6 SD or above +6 SD, with weight-for-age z-scores below -6 SD or above +5 SD, or with weight for height z-scores below -5 SD or above +5 SD are marked as invalid data, following the guidelines of demographic and health surveys (DHS).

Review of Income and Wealth, Series 65, Number S1, November 2019

$$WHZ = \begin{cases} 0 & \text{if healthy} \\ 1 & \text{if moderately wasted} \\ 2 & \text{if severely wasted} \end{cases}$$
$$WAZ = \begin{cases} 0 & \text{if healthy} \\ 1 & \text{if moderately underweight} \\ 2 & \text{if severely underweight} \end{cases}$$

We also define children's nutritional status as binary variables: healthy if $-2 \le z$ -score $\le +2$ and malnourished if z-score ≤ -2 . Hence,

$$haz = \begin{cases} 1 & \text{if stunted} \\ 0 & \text{if healthy} \end{cases}$$
$$whz = \begin{cases} 1 & \text{if wasted} \\ 0 & \text{if healthy} \end{cases}$$
$$waz = \begin{cases} 1 & \text{if underweight} \\ 0 & \text{if healthy} \end{cases}$$

As indicator of maternal malnutrition, we use maternal BMI status with the following categorization: $18.5 \le BMI < 25$ classified as healthy weight, BMI < 18.5 classified as underweight/thin (chronic energy-deficient); and BMI ≥ 25 classified as overweight/obese.⁴ The following binary indicators *m_under* and *m_over* define maternal underweight and overweight status, respectively, as:

$$m_under = \begin{cases} 1 & \text{if underweight} \\ 0 & \text{if healthy} \end{cases}$$
$$m_over = \begin{cases} 1 & \text{if overweight} \\ 0 & \text{if healthy} \end{cases}$$

⁴This category includes women who are obese (BMI \ge 30). In the current sample, only a small fraction (2 percent) of women were obese. The BMI cut-offs are defined for non-pregnant, non-postpatrum women aged 15-49 years. Women whose calculated BMI is below 12.0 or above 60.0 are flagged as out of range and hence excluded from the analysis, following DHS guidelines.

Key Independent Variable: Household Food Security Status (HFS)

The key explanatory variable of interest in our analysis is HFS yielded by the 9-item HFIAS. HFIAS classifies households into four categories of food security status (by access): food secure, mildly food insecure, moderately food insecure and severely food insecure. The mothers were asked nine questions (see Appendix, Table A1) related to the households' experience of food insecurity in the last 30 days preceding the survey. These questions capture three key domains of household food security: 1) anxiety and uncertainty about access (Q1: worried); 2) insufficient quality (Q2-Q4: preferred, variety, nochoice); and 3) insufficient quantity (Q5-Q9: smaller, fewer, nofood, hungry, davnight). Each item starts with an occurrence question that identifies if the condition has been experienced in the household. An affirmative answer is then followed by a frequency-of-occurrence question to determine if the condition happened rarely (once or twice), sometimes (3-10 times), or often (>10 times) during the reference period. The responses are coded as 0 = never, 1 = rarely, 2 = sometimes, or 3 = often. Households that rarely experience some anxiety over sufficiency of food are categorized as food secure. Households that worry about not having enough food frequently as well as households that sometimes in last one month could not have their preferred food or had to eat limited variety of food or food that they really did not want to eat are categorized as mildly food insecure. Households that frequently ate food of limited choice and sometimes had to eat lesser quantity of food are categorized as moderately food insecure. Households that had no food to eat or had to starve day and night are categorized as severely food insecure.

We create categorical variables capturing household food security status based on these scores. The findings from the extant literature indicate that the categorization is important. For example, there exists substantial evidence in the literature that mild or marginal food insecurity can be more detrimental to physical outcomes as opposed to severe food insecurity (Townsend *et al.*, 2001; Cook *et al.*, 2013). Research in the context of developing countries also find evidence of differential effects of various categories of food insecurity on anthropometric measures of nutritional outcomes (Velásquez-Melendez *et al.*, 2011).

Accordingly, variable *fsec3* is defined as an ordered categorical variable representing three categories of food security status– food secure, moderately food insecure and severely food insecure. We collapse mild and moderate food insecurity into one single category *moderately food insecure* to obtain more meaningful results since only 12 percent households are present in each of these categories. However, we retain *severely food insecure* as a separate category since these are the most venerable households and should be identified as such for policy purposes.

Hence *fsec3* is defined as:

$$fsec3 = \begin{cases} 0 & \text{if highly food secure} \\ 1 & \text{if moderately food insecure} \\ 2 & \text{if severely food insecure} \end{cases}$$

We also define HFS as a binary variable by collapsing moderately food insecure and severely food insecure households into one single category *food insecure*:

^{© 2019} International Association for Research in Income and Wealth

$$fsec = \begin{cases} 1 & \text{if food insecure} \\ 0 & \text{otherwise} \end{cases}$$

Table 1 reports the distribution of households by experience of food insecurity. We find, 58 percent of the households never worry about having insufficient food, a little over one-third of the households cannot eat preferred food, 31 percent of the households have to eat only limited variety of food and about a quarter of households have very limited choice of food. About 12 percent of households report that they had to eat smaller meals or had to cut meal size in the last one month though it happened rarely. The extreme situation of not having any food to eat was experienced by 12 percent of the households among which eight percent had this experience rarely. Eight percent of the households reported sleeping hungry without food and only five per cent reported having to go hungry without food day and night.

Control Variables

In modelling the relationship between child nutritional status and HFS, the control variables included in the equations for child nutritional status are: household characteristics such as household size and wealth status; age, gender, education and caste of household head; household environmental conditions captured by variables such as whether the house is pucca or not, whether the household has toilet facility, whether the household has access to piped water and whether the household is located in rural or urban setting; maternal characteristics including maternal educational status, age at marriage, nutritional status captured by BMI and stature (if height <145 cm.), whether mother received at least 3 antenatal care (ANC), maternal work status, mothers' exposure to media and; an empowerment index constructed by collapsing each of the eight survey questions on maternal decision making into binary format (=1 if the mother takes the decision jointly or alone; 0 otherwise) and adding the responses.⁵ The total score on decision-making is a proxy for women's bargaining power in the household (Bloom et al., 2001; Smith et al., 2003; Doss, 2013; Antman, 2014). We also include selected child characteristics as control variables: age, gender, birth order, whether the child was low birth weight (LBW), whether the child was age appropriately breastfed, whether the child consumed at least 4 food groups and whether the child received full immunization. Regional dummies (Amaravati, Konkan, Nagpur, Nasik and Pune) are also included in all models.

Models on maternal nutritional status include the above household level variables and maternal characteristics. However, maternal BMI is now dropped from the models. In the models on maternal undernutrition (indicated by

⁵This methodology of constructing the unweighted index of empowerment is rather simplistic and ideally one should undertake a factor analysis (as in Smith *et al.*, 2003) to combine relevant items. It is also important to weight the decisions. However, the literature has some evidence on the use of similar methodology (see Bloom *et al.* 2001, for example). We examined internal validity of the score by estimating scale reliability coefficient Cronbach's alpha (Cronbach, 1951) which is 0.88 in our case, indicating good reliability (a value of Cronbach's alpha closer to 1 implies greater reliability).

^{© 2019} International Association for Research in Income and Wealth

Percent Distribution of Households by Frequ	UENCY OF EXPI	ERIENCE OF FO	od Insecurity (usi)	NG HFIAS), N	A AHARASHTRA	, 2012
Food Insecurity Exnerience in One Month Prior to	Fi	requency of	Experience of Fo	od Insecuri	ty	Number of
Burvey	Never	Rarely	Sometimes	Often	Total	Households
Worry about insufficient food (worry)	57.9	24.2	14.5	3.4	100.0	2,624
[nability to eat preferred food (preferred)	63.4	21.0	14.0	1.6	100.0	2,617
Had to eat limited variety of food (variety)	68.6	17.6	11.8	2.1	100.0	2,624
Had to eat certain food items without choice (nochoice)	74.5	15.5	9.0	1.0	100.0	2,628
Had to eat smaller meals (smaller)	79.7	12.3	7.6	0.4	100.0	2,627
Had to eat fewer meals (fewer)	81.8	11.2	6.6	0.3	100.0	2,627
Had no food to eat (nofood)	87.8	7.6	4.3	0.3	100.0	2,626
Had to sleep without food (hungry)	92.3	5.2	2.4	0.0	100.0	2,626
Had to go day and night without eating any food	95.5	3.0	1.5	0.0	100.0	2,627
(daynight)						

TABLE 1

underweight status) we include additional maternal characteristics such as number of times a mother was pregnant, status of iron and folic acid consumption and frequency of consumption of seasonal fruits and dark green leafy vegetables during the last 30 days.⁶ Models on maternal overweight include frequency of consumption of oilseeds as an indicator of unhealthy food consumption.⁷

In addition to controlling for maternal empowerment and the household demographics included in the equations for nutritional status, the food security equation includes additional household level covariates such as the type of ration card possessed by the household (no card, yellow, white or orange)⁸, household head's home ownership status and land ownership status and whether the household has some transport.

2.4. Summary Statistics

Table 2 reports summary statistics of variables. We find, 57 percent of households are food secure with higher fraction reported for urban areas. Twenty-three percent of children are stunted, while 16 percent are wasted and about 23 percent are underweight. Prevalence of LBW is similar in both rural and urban areas. One in every three non-pregnant mothers is underweight, as opposed to 11 percent non-pregnant mothers who are either overweight or obese. A higher proportion of underweight mothers is reported in rural areas, while a very high proportion of overweight mothers is noted in urban areas. Nearly 41 percent of the urban population belong to the highest wealth quintile as opposed to only two percent in the rural areas. Twenty percent of households are headed by persons who do not read or write. Only five percent of the households are female headed. Sixtysix percent of the households have access to piped drinking water, the proportion being much higher in urban areas. Fifty-five percent of households do not have access to any toilet facility. About one-fourth of the households may not have access to public distribution system (PDS) as they do not possess ration cards. In rural areas, the proportion of households having a yellow card is more than double of that in urban areas. Nearly one-third of the mothers were married before the legal age of 18 years which makes them vulnerable to early pregnancies. Twelve percent of mothers have received no formal education and majority of them (69 percent) are not engaged in any work besides household chores. Regarding the eight questions on maternal decision-making power, more than 60 percent of women take each decision jointly or alone; with majority (more than 75 percent) participating in decisions to spend their own money, decisions on food items given to the child and decisions on child immunization. Accordingly, the average empowerment score is quite high (5.61, SD 0.09) with a higher mean

⁶Fruits and dark green leafy vegetables are reliable indicators of diet quality (micronutrient intake), fruits being the one most promoted by the nutritionists (Binkley and Golub, 2011).

⁷A rapid increase in consumption of low-cost vegetable oil during the last few decades has been a major component of food system changes and consequent effects on body mass index in terms of obesity (Popkin, 2007).

⁽⁶⁾ Three card categories are issued: Yellow: families having annual income up to Rs. 15,000/; Orange/ Saffron: families having total annual income of more than Rs. 15,000 and less than 1 lakh; White: above poverty line - the families having annual income of Rs. 1 lakh or above (see details in: http://mahafood. gov.in/website/english/PDS.aspx.)

in urban areas. More than 50 percent of mothers in rural Maharashtra never or only occasionally consumed seasonal fruits in the last 30 months as opposed to 40 percent in urban areas. However, about 60 percent of mothers reported consuming oilseeds daily or on alternate days, with higher proportion reported for urban areas.

2.5. Conceptual Framework

In this section, we discuss some key conceptual issues related to our variables of interest before proceeding to building and estimating the empirical models.

The conceptual framework for our empirical models is based on the UNICEF model (UNICEF, 1990) (Figure B.1) as mentioned earlier in the introductory section. The theoretical perspective is provided by Sen (1985, 1999) who refers to the conversion factors and rates in explaining the relationship between food access, food intake and nutritional achievement–conversion of personal income, resources and commodities into well-being which will depend "crucially on a number of contingent circumstances, both personal and social" (Sen, 1999, p. 70): personal heterogeneities such as body size, nutrient requirements or intelligence; environmental conditions such as having access to proper sanitation and safe drinking water, access to transport; variations in social climate such as societal hierarchies; power relations related to class, gender, race, or caste; or differences in relational perspectives and distribution within the family (Burchi and De Muro, 2016). The choice of control variables in our empirical models follows the above conceptual and theoretical paradigms.

The food insecurity-malnutrition relationship has concerns with potential endogeneity of experiential food insecurity possibly arising from three sources. First, since health endowment and food insecurity are difficult to observe the variables may be simultaneously determined due to the presence of unobserved confounding factors affecting both outcomes. Examples of such unobserved variables might be "ability" which may affect both outcomes via the channel of educational attainment.9 Additionally, individuals' nutritional intake and needs are very different, and it is often difficult to measure "needs" which could be related to several endogenous factors such as activity levels, body size and previous nutrient intakes (Strauss et al., 2000). Furthermore, as mentioned previously, there may exist feedback effect from malnutrition to food insecurity (Burchi and De Muro 2016) via the channels of poor nutrition-low productivity-low earnings-poverty. A large body of growing literature also indicates that the health-hunger relationship may be bidirectional (Tarasuk et al., 2013; Anderson et al., 2016), while poor health in turn is often a consequence of malnutrition (Black et al., 2008). A third source of endogeneity could be random measurement error in anthropometric as well as experiential food security indicators (Diskin, 1994). We exploit the sound empirical framework offered by recursive bivariate models¹⁰ (Greene, 2012) (see the next section on Statistical Analysis) to address the above concerns.

⁹Theoretically, better educational outcomes should expand people's basic capability which in turn should enable them to be well nourished and food secure in the long run.

¹⁰See Costa-Font and Jofre-Bonet (2008) for an application.

 $^{{\}ensuremath{\mathbb C}}$ 2019 International Association for Research in Income and Wealth

Next, we provide an understanding about the underlying theoretical framework of HFIAS. The construction of the EBFSS rests on non-linear factor analytic models such as the Rasch model (Bond and Fox, 2001). These models are located within the broad spectrum of Item Response Theory commonly employed in the education literature to conduct psychometric assessment of individuals' ability (latent variable) to answer progressively more difficult questions. In the food security literature, the latent variable is food insecurity and the items representing the underlying construct of interest are arranged along a continuum of severity. Under the above modelling framework food insecurity is viewed as a continuous, unidimensional and unobservable quantity that varies from household to household. Guided by the principles of Rasch model,¹¹ the selection of items in the scale is implemented by undertaking appropriate statistical tests on scale reliability and validity (see Hamilton *et al.*, 1997 for details). The scale items may broadly have universal applicability in diverse settings, however, for local adaptation the validity and reliability tests are recommended to ensure unbiased and reliable results.¹²

At this stage, we would like to highlight some concerns relating to the temporal dimensions of the food security and nutrition indicators. Inherently, food insecurity is a dynamic concept and embedded in it is the element of "time" (Maxwell and Smith, 1992). Issues relating to transient versus chronic food insecurity are likely to have differential impact on nutritional outcomes (Hernandez and Jacknowitz, 2009). Households transition in and out of food insecurity due to its cyclical/seasonal nature or due to appearance of sudden shocks such as price hike (Maxwell and Smith, 1992). Interestingly, a household identified as currently food insecure may also have reached that status by choice-going hungry now in order to prevent selling assets and going hungry later (De Waal, 1989). Therefore, when the EBFSS are used for analytical purposes in cross section framework, they may not be able to differentiate between transitory and chronic food insecurity (Devereux, 2006). While longitudinal data could be a potential solution, a major shortcoming of this literature is limited availability of such data, especially in the context of developing countries. However, even cross-sectional analysis is useful as different points in the life- course reveal a different picture of how food insecurity is related to nutritional outcomes (NRCIM, 2013). Additionally, the two time periods addressed by the survey items in the experiential food security scales (12 months and 30 days) and the follow-up items regarding frequency of occurrence are designed to capture some aspects of periodicity of food insecurity (Hamilton et al., 1997). For example, hunger, the more severe form of food insecurity may occur at the end of the month when household resources are depleted but may disappear when payments or food assistance is received (Hamilton et al., 1997). Similar ideas are reflected

¹¹In general, the dichotomous Rasch model is used to analyze dichotomous items (e.g. yes/no) data, whereas the partial credit model developed by Masters (1982) is used as an extension of the dichotomous Rasch model for analyzing polytomous response pattern (never/rarely/sometimes/often) as in HFIAS.

¹²In our case, we undertook this exercise guided by Dr. Mark Nord from the Voices of the Hungry Project, FAO. Some concerns were detected with the use of two of the scale items (*smaller* and *fewer*). Accordingly, we re-estimated the models by dropping one of the items. However, none of the key conclusions changed substantially. Therefore, in this paper we have presented results based on the original nine-item scale. The details of the results on scale validity and reliability are not reported here but are available upon request.

in Devereux (2006) who argues, in the context of self-assessment indicators, that the recall component "would at least allow a crude categorization to be made of a household as either food secure, chronically food secure or transitorily food secure over a defined time period" (Devereux, 2006, p. 40).

A related issue is that often the anthropometric indicators (such as stunting) indicate long run nutritional deprivation. Therefore, statistical analysis based on indicators such as the EBFSS and stunting in cross sectional framework may seem apparently conflicting. We partially reconcile this contradiction by reiterating that some temporality is captured by the EBFSS via the recall component and the occurrence questions. Quantity reduction (for example, nofood) which occurs "often," reflects hunger and hence may be aligned with chronic nutritional deprivation since hunger and malnutrition are closely related concepts (see Hamilton et al., 1997). Anthropometric indicators of malnutrition in fact represent differential temporal dimensions. Wasting, for example, is important for the description of current health status. Weight-for-age may not be able to discriminate between short and long-term forms of nutritional deprivation given that children classified on its basis are a composite group in terms of their nutritional status – stunted or wasted or both (Fishman et al., 2004). Following Devereux (2006), "to some extent the anthropometric indicator of height-for-age can be used as a proxy for chronic food insecurity... The anthropometric indicator of weight-for-height which measures recent weight loss (wasting) and is caused by inadequate current food intake ... can be used as a crude proxy for transitory food insecurity" (Devereux, 2006, p. 18). The interconnectedness of chronic and transitory food insecurity is also relevant in this context-recurring episodes of transient food insecurity create chronic food insecurity conditions (Maxwell and Smith, 1992). At this point, it may be worthwhile to point out that the literature has evidence that temporary food insecurity may have greater adverse nutritional consequences than persistent food insecurity (Hernandez and Jacknowitz, 2009).

3. STATISTICAL ANALYSIS

3.1. HFS and Maternal and Child Nutritional Status

We test our key hypothesis on the association of HFS with maternal and child nutritional status using the following empirical strategy: i) first we examine bivariate association of HFS and anthropometric indicators; ii) next we estimate multivariate association of nutritional outcomes and HFS using baseline models controlling for selected household, maternal and child characteristics: ordered probit (Oprobit) models for child nutritional status and probit models for maternal nutritional outcomes; and iii) at the final stage we modify our baseline models to address the issue of endogeneity of HFS with nutritional indicators. As mentioned previously, we use the framework of simultaneous equation models—recursive bivariate ordered probit (RBOprobit) & recursive bivariate probit (RBprobit) models (Greene, 2012)—to model the HFS-nutrition linkage. All models are estimated in stata accounting for survey design.

Two additional investigations are undertaken at this stage to ensure robustness. First, we add an interaction term on HFS and the empowerment score

^{© 2019} International Association for Research in Income and Wealth

(*fsec*em_power*) in the RBprobit models.¹³ This modification would allow us to examine whether the effect of HFS on anthropometric indicators changes with increase in women's decision-making power, hence shedding some light on the issue of intrahousehold distribution of food. Greater bargaining power (reflected in higher decision-making score) should allow women more control over household resources and hence ensure more equitable distribution of food within the household (Smith *et al.*, 2003). Accordingly, any adverse effect of food insecurity on women's and children's nutritional status should be smaller or may even be eliminated.

The second robustness check relates to examining the effect of each domain of food security captured by HFIAS (uncertainty, inadequate quality and inadequate quantity) on nutritional status.¹⁴ In undertaking this exercise, we collapse each of the nine questions with polytomous response pattern (never, rarely, sometimes, often) to binary response variables: =1 if sometimes /often; =0 if never or rarely. We subsequently exploit the RBprobit framework to examine these effects by substituting the binary food insecurity variable (*fsec*) by each of the nine binary responses. We estimate the models without the interaction term but retaining all other original control variables.¹⁵

3.2. HFS and Child Nutritional Status

Ordered Probit Model

Baseline Oprobit models are estimated for three categories of child nutritional status—stunting (HAZ), wasting (WHZ) and underweight (WAZ). Three categories of HFS represented by the variable *fsec3* is the key explanatory variable of interest in each equation.

The Oprobit model for each indicator of child nutritional status is specified in the following manner. Let the latent nutritional status be denoted by HAZ^* , WHZ^* and WAZ^* ; respectively for stunting, wasting and underweight. The underlying model for stunting consists of an equation relating the latent nutritional status HAZ^* and food security status *fsec3** to background characteristics represented by vector x_1 (described in Table 2) in the following manner:

(1) $HAZ_{i}^{*} = \gamma fsec3_{i}^{*} + x_{1i}\beta_{1} + \varepsilon_{1i}$

¹³We could only examine this issue in the context of the RBprobit models because HFS is an ordered variable in the RBOprobit models which renders the use of interaction terms and calculation of marginal effects rather challenging.

¹⁴It must be noted that this exercise is somewhat rudimentary since by unidimensionality assumption of Rach model, the nine items should collectively identify the underlying latent trait of food insecurity.

¹⁵We restricted this exercise to RBprobit models assuming that the bivariate framework would be adequate to give us some insight into the nature of this association. The original RBprobit model specifications are retained on the assumption that they would not change substantially even if we substitute the composite food security indicator by an individual dimension of the scale. However, the interaction term is excluded from the model because this exercise is part of an independent robustness check.

Variables	Description	Total	Rural	Urban
Child anthropometric indicators Ordered Categorical				
HAZ	=0 if healthy =1 if moderately stunted	0.772 0.153	0.752 0.166	$\begin{array}{c} 0.797 \\ 0.130 \\ 0.130 \end{array}$
ZHM	=2 if severely stunted =0 if healthy =1 if moderately wasted	0.075 0.836 0.118	0.078 0.829 0.123	$\begin{array}{c} 0.070\\ 0.845\\ 0.112\\ 0.12\end{array}$
WAZ	=2 it severely wasted =0 if healthy =1 if moderately underweight =2 if severely underweight	0.046 0.774 0.160 0.065	0.048 0.750 0.182 0.066	$\begin{array}{c} 0.043\\ 0.803\\ 0.134\\ 0.063\end{array}$
Binary haz_s whz_w waz_u	=1 if stunted, else 0 =1 if wasted, else 0 =1 if underweight, else 0	0.228 0.163 0.225	$\begin{array}{c} 0.248 \\ 0.171 \\ 0.248 \end{array}$	$\begin{array}{c} 0.203 \\ 0.155 \\ 0.197 \end{array}$
Maternal anthropometric indicators <i>m_under</i> <i>m_over</i>	=1 if underweight, else 0 =1 if overweight/obese, else 0	$0.323 \\ 0.110$	0.409 0.039	$0.220 \\ 0.193$
Ney inaepenaent variaotes fsec3 (categorical) fsec (binary)	=0 if HH is food secure =1 if moderately food insecure =2 if severely food insecure = 1 if HH is food insecure, else 0	0.572 0.294 0.132 0.427	$\begin{array}{c} 0.502 \\ 0.322 \\ 0.175 \\ 0.498 \end{array}$	$\begin{array}{c} 0.659\\ 0.259\\ 0.080\\ 0.340\end{array}$
HH socio-economic characteristics hilsize genderhead headlit	household size =1 if female headed, else 0 =0, if HH head does not read/write =1 if HH head has below primary level education =2 if HH head has colleos/university level education =3 if HH head has colleos/university level education	6.4 (0.08) 0.048 0.196 0.144 0.494 0.165	6.6 (0.12) 0.028 0.242 0.165 0.492 0.100	6.3 (0.12) 0.074 0.140 0.117 0.496 0.245

TABLE 2 Summary Statistics Of Variables, Maharashtra, 2012

 ${\ensuremath{\mathbb C}}$ 2019 International Association for Research in Income and Wealth

	TABLE 2 Continued			
Variables	Description	Total	Rural	Urban
headage caste	age of HH head =0 if scheduled caste, omitted base group =1 if scheduled tribe =2 if other backward class =3 if others	66.2 (0.02) 0.153 0.249 0.2493 0.348	72.3 (0.35) 0.124 0.350 0.213 0.313	58.8(0.03) 0.184 0.132 0.290 0.395
$RPDS^{a}$	=0 if, HH has white card =1 if HH has no ration card =2 if orange card =3 if yellow card	Total 0.048 0.238 0.448	Rural 0.028 0.367 0.367	Urban 0.072 0.236 0.543
wealth	 1 if second wealth quintile 2 if middle wealth quintile 3 if fourth wealth quintile 4 if highest wealth quintile 1 if household head owns clear 0 	0.200 0.2048 0.1964 0.2029 0.1956	0.345 0.321 0.081 0.081	0.028 0.056 0.158 0.351 0.411
ownhome ^a ownland ^a transport ^a	=1 if household head owns land, else 0 =1 if at least some, else 0	0.872 0.450 0.509	0.515 0.679 0.515	0.773 0.169 0.501
HH environmental characteristics hhtype: piped toilet	=1 if kachha/semi-pucca , else 0 (pucca)=1 if HH does not have access to piped water, else 0=0 if no school	0.372 0.666 0.554	0.223 0.486 0.335	$\begin{array}{c} 0.552 \\ 0.887 \\ 0.823 \end{array}$
Maternal characteristics mschool	 =1 if primary =2 if secondary to middle, else 0 =3 if above secondary =1 if mother's age at marriage <18, else 0 =1 if mother decides HH purchase of regular food 	0.116 0.058 0.614 0.211	0.143 0.076 0.663 0.117	0.082 0.035 0.553 0.328
agemarriage	litems alone or jointly, else 0 =1 if mother decides food items to be given to the	0.297	0.367	0.210
foodpurchase	child jointly or alone, else 0 =1 if mother decides HH purchase of regular food	0.631	0.605	0.662
decisionfood	items alone or jointly, else 0 =1 if mother decides food items to be given to the child jointly or alone, else 0	0.778	0.783	0.772

Review of Income and Wealth, Series 65, Number S1, November 2019

Variables	Description	Total	Rural	Urban
decisionmoney	=1 if mother decides how to spend her own earnings	0.773	0.772	0.779
decisionhealth	jointly or alone, else 0 =1 if mother takes decision on her own health care	0.671	0.674	0.667
decisionvisit	jointly or alone, else 0 =1 if mother decides visiting family or relatives jointly	0.672	0.666	0.680
hhpurchase	or alone, else 0 =1 if mother decides major household purchase jointly	0.657	0.655	0.661
decisionbaby	or alone, else 0 =1 if mother decides place of delivery of baby jointly	0.649	0.625	0.679
decision_immune	or alone, else 0 =1 if mother decides child immunization jointly or	0.797	0.814	0.777
em_power ^b	alone, else 0 empowerment index	5.61 (0.089)	5.59 (0.115)	5.64
lowheight	=1 if mother's height <145 cm, else 0	10.6	11.0	(0.141) 10.2
ANC	=1 if mother received at least 3 ANC, else 0 =1 if mother did not consume iron folic acid (IFA)	Total 0.906	Rural 0.867	Urban 0.954
ironfolic ^d	more than 90 days, else 0 =1 if mother does not have media, (print/electronic)	0.567	53.2	6.09
media work weenen.ve	exposure, else 0 = 1 if mother has not worked in last 12 months, else 0 Number of times a mother was pregnant =0 if consumption of dark original party voortables is	0.253 0.686 2.11/118)	0.363 0.535 2.15.01180	0.117 0.871 2.08
darkgreen ^d	daily/alternate days	72.4	65.7	(1.18) 79.2
b2	=2 if never/occasionally =0 if fruit consumption is daily/alternate days	20.7 6.4	26.1 7.7	15.3 4.9 32.4
Prutts	=1 II weekly =2 if never/occasionally	26.5	13.9 26.3	32.4 26.7
oil^{f}	=0 if oilseed consumption is daily/alternate days =1 if weekly	50.2 60.1	58.9 54.2	39.6 65.9
	=2 if never/occasionally	3.1 36.5	2.7 43	3.9 30

	TABLE 2 Continued			
Variables	Description	Total	Rural	Urban
Child characteristics childage	age of child in completed months =1 if girl, else 0	11.3 (0.17)	11.1(0.25)	11.5
childegender LBW (low birth weight) ^c	=1 if LBW, else 0 =1 if first order, else 0 (higher order)	0.447 0.201	0.439 0.210	$0.24 \\ 0.458 \\ 0.194$
birthorder BF_age_appr Seven_For_recoded	=1 if first order, else 0 (higher order) =1 if child age-appropriately breastfed, else 0 =1 if child received food from >4 food groups	0.431 0.731 0.088	0.409 0.745 0.091	0.447 0.709 0.085
setting	=1 if child received full immunization, else 0 (partial) =1 if urban, else 0 (rural)	0.371	0.340	0.409
region	=0 if Amaravatı, omitted base group =1 if Aurangabad = 2 if Konkan	$0.10 \\ 0.17 \\ 0.25$		
	=3 if Nagpur =4 if Nasik =5 if Pune	0.08 0.17 0.21		
^a Included in food security equations a ^b Empowerment score is constructed b 0-8.	lone. y adding the responses to the eight binary decision-making questio	ons. Therefore, rang	e of values for th	e score will be
^c Included in equations for stunting alc ^d Included in the equation for materna ^e Included in equations for maternal un ^f Included in the equation for maternal <i>Source</i> : CNSM 2012 (IIPS-UNICEF, 2	ne. l undernutrition alone. nder- and overweight. overweight alone. (013) & Author's compilation.			

Review of Income and Wealth, Series 65, Number S1, November 2019

^{© 2019} International Association for Research in Income and Wealth

where $i = 1, 2, ..., n, \beta_1$ is a column vector of unknown parameters, γ is an unknown scalar which measures the effect of HFS on child nutritional status, ε_{1i} is the error term assumed to be distributed standard normal.

Hence, HAZ^* has an observable counterpart HAZ generated by the observability condition:

$$HAZ = \begin{cases} 0 & \text{if } \mu_{-1} < HAZ^* \le \mu_0 \\ 1 & \text{if } \mu_0 < HAZ^* \le \mu_1 \\ 2 & \text{if } \mu_1 < HAZ^* \le \mu_2 \end{cases}$$

where $\mu_i s$ are the threshold parameters. The variable *fsec3** is generated by the observability condition:

$$fsec3 = \begin{cases} 0 & \text{if } \mu_{-1}' < fsec3^* \le \mu_{0}' \\ 1 & \text{if } \mu_{0}' < fsec3^* \le \mu_{1}' \\ 2 & \text{if } \mu_{1}' < fsec3^* \le \mu_{2}' \end{cases}$$

where $\mu_i's$ indicate thresholds as above.

Similarly, the equations for wasting and underweight are:

(2)
$$WHZ_{i}^{*} = \delta fsec3_{i}^{*} + x_{2i}\beta_{2} + \varepsilon_{2i}$$

(3)
$$WAZ_i^* = \theta fsec3_i^* + x_{3i}\beta_3 + \varepsilon_{3i}$$

Recursive Bivariate Ordered Probit Model

We estimate RBOprobit models with three categories of child nutritional status (indicated by HAZ, WHZ and WAZ) and HFS (*fsec3*) as joint dependent variables, with the latter appearing as an ordinary pre-determined variable on the right-hand side (RHS) of the nutrition equation. The model belongs to a general class of simultaneous equation models discussed by Heckman (1978), Maddala (1983) and Greene (2012). What makes it recursive is the fact that the potentially endogenous explanatory variable *fsec3* appears as pre-determined variable on the RHS of equations for the anthropometric indicators, however, HAZ or WHZ or WAZ do not appear on the RHS of the equation for *fsec3* (Greene, 2012).

We elaborate below the model specifications for stunting (HAZ). Specifications for WHZ and WAZ would be similar. The underlying model for stunting consists of two separate equations relating HAZ^* and $fsec3^*$ to background characteristics represented by vectors x_1 and x_0 , respectively.

(4)
$$fsec3_i^* = \beta_0 x_{0i} + \epsilon_{0i}$$

(5)
$$HAZ_i^* = \gamma' fsec3_i + \beta'_1 x_{1i} + \varepsilon'_{1i}$$

 ${\ensuremath{\mathbb C}}$ 2019 International Association for Research in Income and Wealth

where β_0 and β'_1 are column vectors of unknown parameters, γ' is the unknown scalar which measures the effect of HFS on nutritional status, ε_{0i} and ε'_{1i} are the error terms assumed to be distributed standard normal. The background characteristics listed in x_{0i} and x_{1i} are detailed in Table 2. Full efficiency in estimation and an estimate of γ' are achieved by full information maximum likelihood estimation method (Greene, 2012).

If nutritional status and food security status are jointly determined, estimating the Oprobit equation (Equation 5) in isolation, will give a biased estimate of γ' (Greene, 2012). The possible joint determination of HAZ_i^* and $fsec3^*$ are accounted for by allowing the errors ε_{0i} and ε'_{1i} to be distributed according to a standard bivariate normal distribution with correlation as shown below:

$$E(\varepsilon_{0i}) = E(\varepsilon'_{1i}) = 0$$

$$Var(\varepsilon_{0i}) = Var(\varepsilon'_{1i}) = 1$$

$$Cov(\varepsilon_{0i}, \varepsilon'_{1i}) = \rho$$

The above model allows us to conduct an endogeneity test to check the potential endogeneity of *fsec3* by testing the significance of ' ρ '. The single equation Oprobit model outlined in Equation 1 is a special case of the RBOprobit with $\rho = 0$. If ρ is not significantly different from zero, one concludes that single equation Oprobit may be suitable for estimation purposes. Our coefficient of interest in Equation 5 is γ' . Similarly, our coefficients of interest for *WHZ**, *WAZ** are δ' and θ' , respectively, in the following models:

$$WHZ_i^* = \delta' fsec3_i + \beta'_2 x_{2i} + \varepsilon'_{2i}$$

$$WAZ_i^* = \theta' fsec3_i + \beta'_3 x_{3i} + \varepsilon'_{3i}$$

where x_2 and x_3 are vectors of control variables detailed in Table 2.

Recursive Bivariate Probit Model

For robustness, instead of considering three categories of nutritional status and HFS, we use binary specifications of anthropometric indicators (haz_s , waz_s , whz_s) and HFS (*fsec*). RBprobit models are used to examine the effect of HFS on child nutritional status in this context. The essence of the model is similar to the RBOProbit model outlined in the previous section. In fact, the latter is an extension of the RBprobit model with more than two categories of the dependent variable.

3.3. HFS and Maternal Nutritional Status

We examine the impact of HFS (*fsec*) on maternal underweight (m_under) and overweight (m_over) separately since these indicators reflect different dimensions of nutritional imbalance – under- and overnutrition, respectively. The respective control variables are described in the section on variables (see also

^{© 2019} International Association for Research in Income and Wealth

Table 2). Following the baseline probit models, recursive bivariate probit models are estimated as above.

Identification

Both sets of models (RBOprobit and RBprobit) are estimated by imposing exclusion restrictions even though it is not strictly necessary. "Identification by functional form" is possible which only requires variations in the set of exogenous regressors (Wilde, 2000).¹⁶ One of the identifying variables in the food security equation is *RPDS* denoting the type of ration card possessed by the household. This variable is not included in the nutritional status equation since having a ration card may have a direct impact on HFS (Gopichandran et al., 2010) but not on anthropometric indicators.

4. **R**ESULTS

4.1. HFS and Child Nutritional Status

In Table 3, unadjusted models report that HFS is positively and significantly associated with stunting (rho 0.16, p < 0.05); while the association is weakly significant with respect to wasting (rho 0.18, p < 0.08) and statistically insignificant with respect to underweight. Driven by this mixed evidence, we now turn to the results of the fully adjusted models which control for potential predictors of child nutritional status.

Table 4 reports marginal effects of HFS on child nutritional status for various specification of the models.¹⁷ In reporting the results we focus, primarily, on the RBOprobit/RBprobit models. The results of the baselines Oprobit models are reported for the purpose of comparison only.¹⁸

Results change noticeably in fully adjusted multivariate models. Stunting (HAZ) is no longer significant when potential predictors are adjusted for in the single equation Oprobit model. Moderate food insecurity increases the risk of moderate level of underweight (WAZ = 1), however, the effect is weakly significant (p < 0.09); while the effects of all categories of food security status on wasting (WHZ) are statistically insignificant.

Results change markedly again in the simultaneous ordered probit models. In the RBOprobit model, severe household food insecurity increases the risk of moderate level of child wasting (WHZ = 1) by 0.09 (p < 0.045). Moderate and severe food insecurity significantly increases the risk of *moderate* underweight (WAZ = I) by 0.08 (p < 0.03) and 0.12 (p < 0.04), respectively; while severe level of underweight (WAZ = 2) is only weakly affected by moderate food insecurity (p < 0.06). The marginal effect of HFS on stunting remains statistically insignificant as before.

¹⁶Since identification by functional form relies heavily on the assumption of bivariate normality, it is common practice to impose exclusion restrictions to improve identification (Roodman, 2011). ¹⁷Estimating marginal effects in RBOprobit models is rather complicated and statistical softwares

such as Stata do not routinely estimate them. To estimate the marginal effects in RBOprobit models we used David Roodman's "cmp" framework in Stata (Roodman, 2011). ¹⁸The complete results of the fully adjusted models are not reported but are available on request.

	Food Secure	Moderately Food Insecure	Severely Food Insecure	Rho ^a
Child nutritional status				
Stunting				
Healthy	0.803	0.739	0.712	0.16**
Moderately Stunted	0.146	0.170	0.145	
Severely stunted	0.051	0.090	0.142	
Underweight				
Healthy	0.810	0.737	0.698	0.18
Moderately Underweight	0.140	0.190	0.185	
Severely underweight	0.049	0.072	0.116	
Wasting				
Healthy	0.844	0.841	0.794	0.08*
Moderately Wasted	0.118	0.116	0.123	
Severely wasted	0.038	0.044	0.084	
Maternal nutritional status				
Underweight	0.338	0.405	0.375	0.07*
Overweight	0.196	0.123	0.078	-0.23

 TABLE 3

 Association of Household Food Security Status with Maternal/Child Nutritional Status (Unadjusted), Maharashtra, 2012

Note: ^aPolychoric correlations reported for children and tetrachoric correlations reported for mothers.

**,*Implies significance at 5% and 10% level of significance, respectively.

In all specifications, ρ is statistically significant hence justifying the importance of using simultaneous equation models rather than single equation ordered probit models for the purpose of modelling the food security-nutrition relationship.

Next, we report the results of RBprobit models for the two specifications: with and without the interaction term *fesec*em_power*. Results align closely except with respect to wasting. The marginal effects on stunting are statistically insignificant in all specifications. In the models without the interaction term, moderate food insecurity increases the risk of child wasting by 0.08 (p < 0.04) and the risk of underweight by 0.16 (p < 0.000). However, in models including the interaction term the effect of *fsec* on wasting is no longer significant. The effect on underweight remains statistically significant but is smaller in magnitude. In all models other than stunting, ρ is weakly significant.

Finally, we report the RBprobit results on the association of each component of the composite food security index with each of the three child nutrition indicators (Table 5). For stunting, marginal effect is positive and significant only for two items—the one capturing the dimension of inadequate quality (*preferred*) and the other capturing the dimension of inadequate quantity (*nofood*); while for wasting items capturing quality (*variety* and *nochoice*) and quantity (*fewer* and *nofood*) dimensions are statistically significant. Underweight is the only nutrition indicator which is positively and significantly related to all components of the composite index except for the most extreme condition *daynight* which reflects hunger.

4.2. HFS and Maternal Nutritional Status

Bivariate estimation of the association of HFS with maternal nutritional indicators in the unadjusted models (Table 3) reports that prevalence of maternal

^{© 2019} International Association for Research in Income and Wealth

Association Of Household F	⁷ 00D Security Status W	TTH MATERNAL/CHILD MULTIVARIA	NUTRITIONAL STATUS, MAHARASHTRA 201 NTE MODELS	12: Marginal Effects In Adjusted
Model	Moderately Food Insecure	Severely Food Insecure	Food Insecure (without interaction term: fsec*em_power	Food Insecure (With Interaction Term: fsec*em_power
Child nutritional status Ordered Probit ^a HAZ (=1) HAZ (=1) HAZ (=1) WAZ (=1) WAZ (=1) WAZ (=0) WHZ (=0) WHZ (=1) WHZ (=1) WHZ (=1) WHZ (=1) WHZ (=2) WHZ (=2) WHZ (=1) HAZ (=1) HAZ (=1) HAZ (=1) WHZ (=1) WHZ (=2) WHZ (=1) WHZ (=2) WHZ (=1) WHZ (=2) WHZ (=1) WHZ (=2) WHZ (=1) WHZ (=2) WHZ (=2) WHZ (=1) WHZ (=2) WHZ (=1) WHZ (=2) WHZ (=1) WHZ (=2) WHZ (=1) WHZ (=2) WHZ (=1) WHZ (=2) WHZ (=2) WHZ (=1) WHZ (=2) WHZ	$\begin{array}{c} -0.02 \ (0.018) \\ 0.0139 \ (0.01) \\ 0.0104 \ (0.008) \\ -0.0230* \ (0.012) \\ 0.0230* \ (0.012) \\ 0.0230* \ (0.012) \\ 0.012 \ (0.012) \\ 0.012 \ (0.012) \\ -0.005 \ (0.006) \\ 0.06 \ (0.012) \\ -0.07 \ (0.05) \\ 0.05 \ (0.03) \\ 0.05 \ (0.03) \\ 0.06 \ (0.03) \\ 0.05 \ (0.03) \\ 0.05 \ (0.03) \\ 0.05 \ (0.03) \\ 0.05 \ (0.03) \\ 0.05 \ (0.03) \\ 0.05 \ (0.03) \\ 0.05 \ (0.03) \\ 0.05 \ (0.03) \\ 0.05 \ (0.05) \\ 0.05 \ (0.05) \\ 0.05 \ (0.05) \\ 0.05 \ (0.05) \\ 0.05 \ (0.05) \\ 0.05 \ (0.05) \\ 0.05 \ (0.05) \\ 0.05 \ (0.05) \\ 0.05 \ (0.05) \\ 0.05 \ (0.05) \\ 0.05 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.055 \ (0.05) \\ 0.055 \ (0.05) \\ 0.055 \ (0.05) \\ 0.055 \ (0.05) \\ 0.055 \ (0.05) \\ 0.055 \ (0.05) \\ 0.055 \ (0.05) \\ 0.055 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \\ 0.065 \ (0.05) \ (0.05) \\ 0.065 \ (0.05) \ ($	$\begin{array}{c} -0.017 & (0.028) \\ 0.0009 & (0.015) \\ 0.007 & (0.012) \\ -0.019 & (0.026) \\ 0.0014 & (0.026) \\ 0.0014 & (0.025) \\ 0.0014 & (0.025) \\ -0.003 & (0.016) \\ -0.003 & (0.016) \\ -0.014 & (0.025) \\ 0.008 & (0.05) \\ 0.07 & (0.05) \\ 0.07 & (0.05) \\ 0.07 & (0.05) \\ 0.07 & (0.05) \\ 0.07 & (0.05) \\ 0.07 & (0.05) \\ 0.07 & (0.05) \\ 0.07 & (0.05) \\ 0.07 & (0.05) \\ 0.05 & (0.05) \\ 0.07 & (0.05) \\ 0.07 & (0.05) \\ 0.07 & (0.05) \\ 0.07 & (0.05) \\ 0.07 & (0.05) \\ 0.07 & (0.05) \\ 0.07 & (0.05) \\ 0.07 & (0.05) \\ 0.05 & (0.05) \\ 0.07 & (0$		
WAZ (=2) Recursive Bivariate Probit ^c	0.06* (0.03)	0.09 (0.06)		
haz whz waz			$\begin{array}{c} 0.04 \ (0.07) \\ 0.08**(0.03) \\ 0.16^{***}(0.03) \end{array}$	-0.02 (0.081) 0.06 (0.049) $0.12^{**} (0.055)$
Maternal nutritional status Probit Model <i>m_under</i> (BMI < 18.5)			х ,	0.02 (0.074)

TABLE 4 Child Nutri

 ${\rm \mathbb{C}}$ 2019 International Association for Research in Income and Wealth

	oderately Food	Severely Food	Food Insecure (without interaction term:	Food Insecure (With Interaction Term:
Model (1)	Insecure (2)	Insecure (3)	fsec*em_power (4)	fsec*em_power (5)
<i>m_over</i> (BMI ≥ 25) Recursive Bivariate Probit ^d				-02 (0.02)
m_under			0.17^{**} (0.07)	0.24^{***} (0.07)
m_over			-0.14(0.38)	-0.11 (0.39)
<i>Note:</i> ******Implies significance (<i>wasting</i>) are ordered categorical varial (<i>wasting</i>) are binary variables: =0 if he	at 1%, 5% and 10% bles: =0 if healthy, = althy & =1 if malnou	 b level, respectively. S e1 if moderately malno arished. m_under & m. 	tandard errors in parenthesis. HAZ ourished & =2 if severely malnourish over are binary variables: =0 if moth	(stunting), WAZ (underweight) & WH2 ed. haz (stunting), waz (underweight), wh er is healthy, =1 if mother is underweight
overweight, respectively. ^a Models include child nutritional s	status as ordered ca	tegorical variables. Co	ontrol variables in the equation for st	unting include household characteristics
age, education, gender and caste of hor	usehold head, hous	ehold size, wealth stat	us, whether the household is pucca, v	whether the household has access to piped
water, whether the household has access in last 12 months. whether mother rece	s to toilet facility; m sived at least 3 ANC	aternal characteristic ∴if maternal height <	s: mother's age of marriage (if <18 yrs 145 cm. mothers' exposure to media.), education level, whether mother worked empowerment index: and child character
istics: child's age, gender, birth order, w	hether child is appr	opriately breastfed, w	hether child received at least 4 food g	roups, whether child had low birth weigh
(LBW) and finally setting (rural/urban) and region (Ama	ravati, Aurangabad, F	conkan, Nasik and Pune). The equa	ions for wasting and underweight do no
include the variables LBW and materna ^b Models include child nutritional s	al stature, but instea status and HFS as c	d include maternal BN ordered categorical vai	Al status and whether the child received riables. In addition to controlling for	ed full immunization. • empowerment and the household demo
graphics included in the equation for m	atritional status as s	tated above, the food s	ecurity equation includes additional	household level factors such as the type o
ration card possessed by the household	l (no card, yellow, w	hite or orange), house	hold home ownership and land owne	rship status & whether the household ha
some transport. «Model includes mutritional status	and food security s	tatus as binarv variab	des Additionally column five renort	s result of B hnrohit models including th
interaction term on empowerment scor	e and HFS.			
^d Models include maternal nutritio	nal status and food	security status as bina	try variables. These models do not in	clude child characteristics. All household
level variables mentioned above are inc	luded. Maternal ch	aracteristics include ac	Iditional covariates such as frequency	/ of consumption of fruits and dark gree
leafy vegetables (in underweight equat overweight equations), and whether mo	ion)/consumption o ther consumed iron	f oilseeds (in overwei and folic acid for mor	ght equation), number of times moth e than 90 days (in underweight equat	ter was pregnant (in both underweight δ ion).

TABLE 4 Continued

 ${\ensuremath{\mathbb C}}$ 2019 International Association for Research in Income and Wealth

underweight increases from food secure (34 percent) to severely food insecure households (38 percent), however the relationship is weakly significant (rho 0.07, p < 0.07). Prevalence of overweight decreases with increasing food insecurity, however the relationship is statistically insignificant.

In the fully adjusted probit models (Table 4), HFS has no effect on maternal underweight or overweight. As before, results change distinctly in the RBprobit models which address potential endogeneity of *fsec* (Table 4). We report results for two specifications of the models again, with and without the interaction term, and find that the results are robust. In the model without the interaction term, increase in the probability of food insecurity significantly increases the risk of maternal underweight by 0.17 (p < 0.02); while the inclusion of the interaction term increases the magnitude of the effect to 0.24 (p < 0.001). The marginal effects on overweight status remains statistically insignificant in both specifications. Rho (ρ) is statistically insignificant for various specification of the models.

With regard to the marginal effects of individual dimensions of food insecurity on the risk of maternal underweight and overweight, all effects are statistically insignificant for overweight status while maternal underweight is significantly affected by the quantity dimensions of food insecurity (*fewer*, *nofood*, *hungry*).

5. DISCUSSION

Evidence on the association of food insecurity with nutritional outcomes is mixed. Certain key issues emerge. First, in examining this relationship it is important to address the concerns with potential endogeneity of HFS with respect to anthropometric indicators. Results of the baseline models change markedly when we account for endogeneity within the framework of simultaneous ordered probit/probit models. Rho is insignificant only in the RBprobit models on child stunting and maternal underweight /overweight which imply that in these cases the relationship may be examined using single equation probit / ordered probit models.¹⁹

Second, household food insecurity is not able to explain child stunting in any of the models. Using this same dataset for Maharashtra, Chandrasekhar *et al.* (2017) do not report any association of HFS with child stunting, wasting or underweight. Similar results have been reported in other populations such as: for children aged 6 to 36 months in Ghana (Saaka and Osman, 2013) and for children aged 6–23 months in Nepal (Osei *et al.*, 2010). However, several studies from low-, lower middle- and upper middle-income countries report strong association of HFS with stunting for under-five children — Motbainor *et al.* (2015) and Reis (2012) to cite a few—which apparently contradicts our conclusion. In such cases, our results may not be comparable primarily due to two reasons. First, the empirical strategy used in our paper attempts to address endogeneity which none of the other studies addressed. Second, our study has a younger age-group specification of 0–23 months. The only study which reports significant association between HFS

¹⁹However, this outcome does not remove the endogeneity of HFS with respect to nutritional outcomes (Greene, 2010).

 $^{{\}ensuremath{\mathbb C}}$ 2019 International Association for Research in Income and Wealth

			Models		
	Child Stunting	Child Wasting	Child Underweight	Maternal Underweight	Maternal Overweight
Worried	0.048 (0.086) 0.074*** (0.017)	0.040 (0.049) 0.015 (0.074)	0.110*** (0.021) 0.087*** (0.028)	-0.005(0.115) 0.059(0.05)	-0.207 (0.296) -0.155 (0.278)
Variety	0.066 (0.107)	0.059** (0.020)	0.117*** (0.022)	-0.013 (0.138)	-0.149(0.271)
Nochoice	0.050(0.156)	$0.065^{**}(0.027)$	0.089^{***} (0.019)	0.069(0.153)	-0.275*(0.150)
Smaller	0.009 (0.195)	0.017 (0.111)	0.102^{***} (0.035)	0.040(0.196)	0.046(0.198)
Fewer	-0.051 (0.486)	0.053^{**} (0.022)	0.060^{**} (0.023)	$0.140^{**}(0.057)$	0.001 (0.397)
Nofood	0.042^{***} (0.012)	0.038^{**} (0.015)	0.036*(0.020)	$0.163^{**}(0.039)$	-0.175(0.095)
Hungry ^a	0.029 (0.076)	0.040(0.046)	0.035^{**} (0.012)	$0.028^{***}(0.005)$	-0.136(0.118)
Dayňight ^b	0.055 (0.090)	0.158	0.132(0.088)	0.047 (0.094)	-0.113(0.140)
Note: Each	l item is a binary variable wi	ith the following response	pattern: =1 if no or rarely;	=0 if sometime or often. All var	iables in the models are as de-

~	e models are as de-
	. All variables in th
	e or often
	0 if sometime
	rarely; =
/	=1 if no oi
)	<i>ote</i> : Each item is a binary variable with the following response pattern: = 1 in Table 4 for the R Bprobit models.
•	scribed

Association of Each Item in HFIAS with Maternal and Child Nutritional Status, Maharashtra, 2012: Marginal Effects in Recursive Bivariate Probit TABLE 5

and stunting for children aged 0-24 months (Saha et al., 2009) uses longitudinal data as opposed to cross sectional data used in our study. Importantly, age related differences are crucial in understanding contradictory evidence from various studies. Biologically, linear growth is a slower process than growth in body mass which implies that a significant degree of stunting takes longer to be visible and may not be evident for some years (WHO, 1986); albeit prevalence of stunting increases over time up to the age of 24-36 months after which it may level off (WHO, 1986). Apart from these biological variances there may also prevail geographical and gender related differences in the process of stunting and wasting (WHO, 1986) which help to reconcile conflicting findings from various studies. Furthermore, stunting reflects longer term deprivation driven more by factors such as maternal nutritional deficiency and consequent low birth weight of the child (Black et al., 2008). Lack of access to improved sanitation and drinking water could be the other major contributors. Incidentally, in 2016, only 52 percent households in Maharashtra used improved sanitation facility which is only marginally higher than the national average (48 percent) and far below the average in Kerala (98 percent); while drinking water coverage has not improved in the state over the last decade 2006–16 (NFHS, 2017). It is also possible that food insecurity observed in Maharashtra is more cyclical or temporary rather than being chronic and hence not associated with stunting (Devereux, 2006). Even with chronic food insecurity, households may develop some coping strategies to deal with food insecurity (Hernandez and Jacknowitz, 2009). In either of the above cases we would not expect any association of HFS with stunting. Among the various components of the composite food security index, the only component which reports any association with child stunting is 'no food to eat of any kind' reflecting severe food insecurity.

Evidence on the impact of HFS on child wasting from various models is mixed. The recursive bivariate specification of the model without the interaction term reports significant increase in the risk of wasting due to HFI. Further disaggregating results, from the recursive ordered specifications we find that it is severe food insecurity which influences wasting. However, the effect of food insecurity on wasting disappears when we incorporate the additional effect of women's bargaining power by including the interaction term in the model. We attempt to reconcile the above findings by positing three sets of arguments. First, the fact that severe food insecurity has an effect on the risk of wasting (moderate) is consistent with the biological process of wasting and its possible link to food insecurity. As discussed previously, wasting captures recent episodes of growth deficit caused by inadequate current food intake and infection possibly due to a sudden a shock (e.g. food price hike) which is typically more acute (Devereux, 2006). Second, as opposed to stunting, wasting is a quicker process (both onset and recovery can be rapid) (WHO, 1986). Additionally, contrary to stunting, prevalence of wasting is greatest between 12 and 24 months of age and tend to decrease later on. These differences help explain why we observe significant association of HFS with wasting but not with stunting for children aged 0-23 months. Regarding why household food insecurity no longer affects child wasting once we incorporate the effect of women's bargaining power (proxied by decision-making), one immediate explanation might be that in times of sudden shocks or acute shortage of food women with higher bargaining power are better able to protect their children from adverse

 $^{{\}ensuremath{\mathbb C}}$ 2019 International Association for Research in Income and Wealth

effects of food insecurity (Quisumbing *et al.*, 1996). The pathways through which improvement in women's status improve child nutritional status may be higher quality care for themselves and their children - better complementary feeding practices, better information about healthy food choices, ability to increase budget share of food and better treatment of illness and immunization of children (Smith *et al.*, 2003). Concerning the alignment of our results with the broader literature, the evidence on food insecurity-wasting link is limited. Only three out of 15 studies reported in Maitra (2018) find positive and significant association of experiential food insecurity with child wasting–for under-five children in Ethiopia (Motbainor *et al.*, 2015), Bangladesh (Hasan *et al.*, 2013) and Nigeria (Ajao *et al.*, 2010). Ali *et al.* (2013) report significant association for under-five children in Bangladesh but not in Ethiopia and Vietnam. The age-related biological factors or geographical factors discussed in the context of stunting may explain the contradictory findings in this context too.

Household food insecurity is a significant driver of underweight in children. Additionally, this finding is robust to various specifications of the models. All individual components of food insecurity captured by the items relating to anxiety, inadequate quality and inadequate quantity also report strong association with child underweight. Interestingly, even when we account for women's decision-making power, the effect of food insecurity on the risk of child underweight remains strong. However, the magnitude of the effect is smaller implying, as before, that greater bargaining power of women promote higher child well-being by dampening the adverse nutritional consequences of food insecurity. Our finding is consistent with such claims from the wider literature that equal status of men and women is likely to reduce prevalence of underweight in children aged under three years in South Asia (Smith et al., 2003). The pathway from food insecurity to child underweight is most likely nutritional, driven by the fact that children from food-insecure households are more likely to consume diet low in energy and micronutrients (Oh and Hong, 2003; Cook et al., 2004; Skalicky et al., 2006). Similar results have been reported for children aged 0-24 months in Bangladesh (Saha et al., 2009); and for pre-school children in Columbia (Hackett et al., 2009).

We find household food insecurity to be a risk factor for maternal underweight as well. As with the case of child underweight, this finding may be explained by low total energy intake and poor nutrient intake caused by inadequate access to food (Tarasuk and Beaton, 1999). Additionally, in the face of household food insecurity mothers often resort to coping strategies by which they compromise their own energy intake to meet their children's need (Isanaka et al., 2007). Particularly, low income mothers may allocate resources or sacrifice their own food consumption to protect children from hunger (Devine et al., 2006). Perhaps this behaviour explains why only the questions related to the 'quantity' dimensions of HFIAS are significantly associated with maternal underweight. Similar findings are reported in Cambodia (McDonald et al., 2015), Nepal (Singh et al., 2014) and Columbia (Isanaka et al., 2007). Interestingly, when we incorporate the effect of women's bargaining power in HFS, the magnitude of the effect of food insecurity on the risk of underweight increases. It is possible that the decision-making variable itself is problematic as a proxy indicator of bargaining power-women may be making decisions within the constraints imposed by their husbands (Doss, 2013). For example,

^{© 2019} International Association for Research in Income and Wealth

they may be given a budget for food and household expenses which they decide how to spend. Such decisions drive better outcomes for the children but may leave themselves deprived once they have taken care of the children. Therefore, the cultural context is also important in interpreting results based on decision- making as proxy to women's bargaining power (Doss, 2013).

We do not find any association of HFS with maternal overweight. This finding is robust across all models, and is also consistent with the broader literature which concludes that any relationship between experiential food insecurity and overweight is limited or absent in developing country settings. Several literature reviews (Nettle et al., 2017; Dinour et al., 2007; Maitra, 2018) conclude that the food insecurity-overweight link in adults is the most pronounced among women in high income countries; and "limited evidence from developing countries may not show the same pattern as the developed countries (especially the USA) from which most of the evidence comes" (Nettle et al., 2017, p. 7). Despite methodological concerns which pervade this literature, the consensus among researchers on this issue is overwhelming. This finding may reflect the fact that in a developing country context household food insecurity might be associated with lack of availability of even the least expensive energy dense food that might lead to overweight. Similar results are reported by Isanaka et al. (2007) for Columbia where adults and children from resource poor households resort to fewer purchase of commercial energy dense food like fried snacks/chips, as opposed to similar population from the US. In the latter case, the individuals from food insecure households facing limited resources consume less expensive and more calorie-dense food to maintain caloric intake at less cost, exposing themselves to greater risk of overweight (Drewnowski and Specter, 2004). In our context, accounting for women's empowerment does not change the conclusion, indicating that the pathways from food insecurity to overweight are likely to be very different from those linking undernutrition and food insecurity (FAO, 2018). Recent discourse on multiple burden of malnutrition and nutrition transition indicate that these pathways may be associated with the particular stage of nutrition transition the country is situated in (see Kac et al., 2012; Doak et al., 2005; Smith, 2015). In the Indian context, labor market inactivity could also be a driver of obesity (Dang et al., 2019). Moreover, overweight and associated dual burden of undernutrition is more prevalent in affluent households in India (Helble and Sato, 2018).

A key message emerging from the above results is, we do have some indirect evidence that food insecurity might be a potential determinant of human development. The strong and robust association of HFS with child and maternal underweight indicates that effects of food insecurity may persist throughout the life cycle. Maternal underweight is a risk factor for low birth weight (Black *et al.*, 2008) with dire consequences such as increased risk of neo-natal mortality and poor child growth both of which are detrimental to formation of human capital (Victora *et al.*, 2008). Low weight-for-age in child has been associated with delayed motor development, poor performance on conservation tasks (Fishman *et al.*, 2004) and lower scores on aggregate measures such as IQ (Aboud and Alemu, 1995) to cite a few. These adverse outcomes affect productivity in the long run, perpetuating the cycle of poverty-food insecurity-poor nutrition-poor learning outcomes-low earnings. Future research should examine the direct effects of food insecurity on

^{© 2019} International Association for Research in Income and Wealth

cognitive outcomes of children and their late life earnings to generate more concrete evidence on the above hypothesis.

6. CONCLUSION

We report mixed evidence with respect to the association of household food insecurity with maternal and child nutritional status. Household food insecurity predicts child wasting and underweight but not child stunting. However, women's bargaining power seems to be dampening the adverse effects of food insecurity on child malnutrition. Furthermore, household food insecurity is a risk factor for maternal underweight but not for overweight. Overall, these findings indicate that policies aimed at improving household food access should improve nutritional outcomes of women and children, however, food alone may not be the solution.

Providing adequate access to food may not address the concerns with child stunting (Rah *et al.*, 2015). Potential solutions may lie elsewhere, such as in the direction of addressing maternal nutritional deficiencies within the notion of first "1000 days" of life (Victora *et al.*, 2008) or in ensuring access to improved drinking water and sanitation (Ngure *et al.*, 2014). Growing evidence suggests a link between child linear growth and household water, sanitation and hygiene (WASH) practices. Thus, a multisectoral approach is warranted for prevention of stunting.²⁰ Dietary diversity is also important in this context highlighting the importance of *quality* (micro-nutrients) of food rather than *quantity* (calories primarily).

Wasting, on the other hand requires actions at the community and household level such as child feeding practices (breastfeeding and complementary feeding), and improved disease environment aimed to reduce infection (for example, promoting intervention on zinc with management of diarrhea). At the community level, programs such as Integrated Child Development Services (ICDS) should be more active in providing adequately nutritious food to the vulnerable population. Quality and timing of school meals must improve which is a concern in the Indian context (Afridi et al., 2016). For example, recently egg has been introduced in ICDS meals which could have higher benefits in terms of improving child nutritional status (Mittal and Meenakshi, 2018). Also, typically cited as a classic example of policy failure (Haddad et al., 1997), often with public food assistance programs such as "school meals" children's food allocation at home is reduced when they receive meals from school. Mittal and Meenakshi (2018) cite evidence of such partial meal substitution at home in the Indian context. Considering wasting is seen as a shortterm response to inadequate dietary intake and is reversible (WHO, 1986), concerted efforts should be made in the above directions. In this context, similar policy recommendations follow for child underweight which is also a likely outcome of inadequate access to food. For example, for children with moderate malnutrition, family food can help recovery as long as the diet is diverse providing adequate

²⁰Accordingly, the UNICEF (1990) conceptual framework has been extended to include WASH interventions (see Haddad *et al.*, 2015).

protein, energy and micronutrients (for example, diet based on legumes, animal protein and dark green leafy vegetables) (Briend *et al.*, 2009).

The strong link between maternal underweight and household food insecurity is concerning because maternal undernutrition has life-long consequences (Victora *et al.*, 2008). The high rates of child undernutrition in south Asia, the so-called "Asian Enigma," has been traced back to higher rates of maternal undernutrition (Ramalingaswami *et al.*, 1996). If thinness is caused by household food insecurity, utmost effort should be directed at providing nutrition-specific interventions such as maternal dietary supplementation or multiple micronutrient fortification (Bryce *et al.*, 2008). The issue of women's status is also important in this context which we discuss below. Interestingly, even if food security interventions may be effective in confronting maternal underweight, different sets of policies may be necessary for dealing with overweight or obesity. In terms of policy, interesting options such as the question of geographical targeting have become relevant—for example, if programs such as maternal spot feeding should offer so many calories to mothers residing in states (such as Karnataka) where geographically overweight prevalence is already high (Sethi, 2018).²¹

A crucial policy implication of the findings of the paper relates to women's empowerment. Even though we have not dealt explicitly with intra-household resource allocation issues in the present paper, our findings indicate that women's bargaining power may have a critical role in mediating the association between household food insecurity and nutritional outcomes of women and children living in the household. Part of the puzzle of 'Asian enigma' is explained by intrahousehold inequity in access to food by women and girls. One straightforward policy implication of this finding is, a greater focus on programs which promote women's empowerment—the well-established nutrition-sensitive programs (Ruel and Alderman, 2013). Examples of such programs are: girls' schooling, conditional cash transfer programs, provision of basic health services or social support schemes to address maternal mental health and concerns with low self-esteem which are risk factors common to both nutrition and child development (Tripathy *et al.*, 2010).

Finally, in the context of policy implications, we refer to the challenges posed by multiple burden of malnutrition (WHO, 2017; Haddad *et al.*, 2015). The Maharashtrian sample does provide some evidence of intra-household dual burden of malnutrition since we observe the presence of both overweight mothers and undernourished (stunted, wasted or underweight) children in our sample. However, since we have not explicitly dealt with such mother-child pairs in the present study, we would not be able to draw any inference on its link to household food insecurity. For our purposes what seems to be important is that the complexities of multiple burden of malnutrition further complicate food and nutrition related policy interventions. For example, if food based polices target households with high overweight prevalence then the underweight individuals residing in dual burden households may become "target of obesity-prevention policies" (Doak *et al.*, 2005). Such possibilities caution us on potential policy failure. These contradictions also take us back to a focus on households rather than individuals at a time when what we

²¹Approximately 23 percent of women in Karnataka were overweight in 2015–16 and the prevalence rate increased from 15 percent in 2006 (NFHS, 2017).

 $^{{\}ensuremath{\mathbb C}}$ 2019 International Association for Research in Income and Wealth

urgently need is a set of holistic policies promoting wellbeing of every individual residing in the household (Doak *et al.*, 2005). Referring to our previous discussion on stunting, policies with an overwhelming focus on *quantity* of food rather than *quality*—"Calorie fundamentalism" (Headey *et al.*, 2012)—are not holistic and are likely to result in persistence of undernutrition and overweight.

Overall, the policy implications of our findings suggest that a combination of nutrition specific and nutrition sensitive policies is required to address the concerns with malnutrition. At this stage we mention some limitations of our research. A major shortcoming of our study is its cross-sectional design. For better understanding of the food insecurity-malnutrition link longitudinal data is essential. The dynamic nature of this relationship would not be fully understood unless we are able to analyze the mechanism of households/individuals moving in and out of food insecurity. Such data also allows us to examine the problem using a life course framework. Second, we could not explicitly model intra-household allocation of resources in the present paper. The proxy indicator capturing women's bargaining power can be improved and information on women's time allocation is crucial (Senauer *et al.*, 1986). At this stage, we recommend further research in this context. Future research should also focus on the rising complexities of multiple burden of malnutrition and related policy issues. The experiential food security indicator also has its limitations (Jones et al., 2013). In general, research should be directed toward exploring better ways of incorporating the time dimension in the experiential scale measures. Furthermore, it is important to be able to measure maternal and child food insecurity at the individual level in order to get reliable and precise estimates of the effect of food insecurity on individual nutritional status. Inferring on child food security status based on parental reporting may be misleading (NRCIM, 2013). Future research should also direct its attention to understanding the various channels from food insecurity to nutritional outcomes for effective policy formulation (FAO, 2018). Such efforts might go a long way in formulating development polices which are "people" centric. Integrating food and nutrition security in policies for human development may be a pathway to inclusive growth, or perhaps more appropriately "nutrition sensitive growth" (Headey, 2011). After all, nutrition may be the bridge between economic and human development (Haddad, 2014).

REFERENCES

- Aboud, F. E. and T. Alemu, "Nutrition, Maternal Responsiveness and Mental Development of Ethiopian Children," Social Science & Medicine, 41, 725–32, 1995.
- Afridi, F., B. Barooah, and R. Somanathan, "The Mixture as Before? Student Responses to the Changing Content of School Meals in India," *Institute for the Study of Labor (IZA)*, 1–41, 2016.
- Ajao, K. O., É. O. Ojofeitimi, A. A. Adebayo, A. O. Fatusi, and O. T. Afolabi, "Influence of Family Size, Household Food Security Status, and Child Care Practices on the Nutritional Status of Under-Five Children in Ile-Ife, Nigeria," *African Journal of Reproductive Health*, 14, 117–26, 2010.
- Ali, D., K. K. Saha, P. H. Nguyen, M. T. Diressie, M. T. Ruel, P. Menon, and R. Rawat, "Household Food Insecurity Is Associated with Higher Child Undernutrition in Bangladesh, Ethiopia, and Vietnam, but the Effect Is Not Mediated by Child Dietary Diversity," *The Journal of Nutrition*, 143, 2015–21, 2013.

- Anderson, P., M. Butcher, F. Kristin, H. W. Hoynes, and D. W. Schanzenbach, "Beyond Income: What Else Predicts Very Low Food Security Among Children?" *Southern Economic Journal*, 82, 1078–105, 2016.
- Antman, F. M., "Spousal Employment and Intra-Household Bargaining Power," Applied Economics Letters, 21, 560–3, 2014.
- Binkley, J. K. and A. Golub, "Consumer Demand for Nutrition Versus Taste in Four Major Food Categories," *Agricultural Economics*, 42, 65–74, 2011.
- Black, R. E., L. H. Allen, Z. A. Bhutta, L. E. Caulfield, M. deOnis, M. Ezzati, C. Mathers, and J. Rivera, "Maternal and Child Undernutrition: Global and Regional Exposures and Health Consequences," *The Lancet*, 371, 243–60, 2008.
- Bloom, S., D. Wypij, and M. Gupta, "Dimensions of Women's Autonomy and the Influence on Maternal Health Care Utilization in a North Indian City," *Demography*, 38, 67–78, 2001.
- Bond, T. G. and C. M. Fox, *Applying the Rasch Model: Fundamental Measurement in the Human Sciences*, Lawrence Erlabaum Associates, Mahway, NJ, 2001.
- Briend, A., Z. W. Prinzo, and A. Briend, "Dietary Management of Moderate Malnutrition: Time for a Change," *Food and Nutrition Bulletin*, 30, S265–S266, 2009.
- Bryce, J., D. Coitinho, I. Darnton-Hill, D. Pelletier, and P. Pinstrup-Andersen, "Maternal and Child Undernutrition: Effective Action at National Level," *The Lancet*, 371, 510–26, 2008.
- Burchi, F. and P. De Muro, "From Food Availability to Nutritional Capabilities: Advancing Food Security Analysis," *Food Policy*, 60, 10–9, 2016.
- Chandrasekhar, S., V. M. Aguayo, V. Krishna, and R. Nair, "Household Food Insecurity and Children's Dietary Diversity and Nutrition in India. Evidence from the Comprehensive Nutrition Survey in Maharashtra," *Maternal & Child Nutrition*, 13(S2), 1–8, 2017.
- Coates, J., A. Swindale, and P. Bilinsky, Household Food Insecurity Access Scale (HFIAS) for Measurement of Household Food Access: Indicator Guide (v. 3), Food and Nutrition Technical Assistance Project, Academy, 2007. Available at: http://www.fao.org/fileadmin/user_upload/ eufao-fsi4dm/doc-training/hfias.pdf.
- Cook, J. T. and D. A. Frank, "Food Security, Poverty, and Human Development in the United States," Annals of the New York Academy of Sciences, 1137, 193–209, 2008.
- Cook, J. T., D. A. Frank, C. Berkowitz, M. M. Black, P. H. Casey, D. B. Cutts, A. F. Meyers, N. Zaldivar, A. Skalicky, S. Levenson, T. Heeren, and M. Nord, "Food Insecurity is Associated With Adverse Health Outcomes Among Human Infants and Toddlers," *Journal of Nutrition*, 134, 1432–8, 2004.
- Cook, J. T., M. Black, M. Chilton, D. Cutts, S. Ettinger de Cuba, T. C. Heeren, R. Rose-Jacobs, M. Sandel, P. H. Casey, S. Coleman, I. Weiss, and D. A. Frank, "Are Food Insecurity's Health Impacts Underestimated in the U.S. Population? Marginal Food Security Also Predicts Adverse Health Outcomes in Young U.S. Children and Mothers," *Advances in Nutrition* (*Bethesda, Md.*), 4, 51–61, 2013.
- Costa-Font, J. and M. Jofre-Bonet, "Is There a 'Secession of the Wealthy'? Private Health Insurance Uptake and National Health System Support," *Bulletin of Economic Research*, 60, 265–87, 2008.
- Cronbach, L., "Coefficient Alpha and the Internal Structure of Tests," *Psychometrika*, 16, 297–334, 1951.
- Dang, A., P. Maitra and N. Menon, "Labor Market Engagement and the Body Mass Index of Working Adults: Evidence from India," *Economics and Human Biology*, 33, 58–77, 2019.
- Devereux, S., Identification of Methods and Tools for Emergency Assessments to Distinguish Between Chronic and Transitory Food Insecurity, and to Evaluate the Effects of the Various Types and Combinations of Shocks on These Different Livelihood Groups, United Nations World Food Programme, Rome, 2006. Available at: https://www.livestock-emergency.net/userfiles/file/ assessment-review/Devereux%202006.pdf.
- De Waal, A., Famine That Kills: Darfur, Sudan, 1984-1985. Clarendon Press, Oxford, 1989.
- Devine, C. M., M. Jastran, J. Jabs, E. Wethington, T. J. Farell, and C. A. Bisogni, "A Lot of Sacrifices: Work-family Spillover and the Food Choice Coping Strategies of Low-Wage Employed Parents," Social Science & Medicine, 63, 2591–603, 2006.
- Dinour, L. M., D. Bergen, and M. C. Yeh, "The Food Insecurity-Obesity Paradox: A Review of the Literature and the Role Food Stamps May Play," *Journal of the American Dietetic Association*, 107, 1952–61, 2007.
- Diskin, P. K., Understanding Linkages Among Food Availability, Access, Consumption, and Nutrition in Africa: Empirical Findings and Issues from the Literature, vol. 54707, Michigan State University, Department of Agricultural, Food, and Resource Economics, Michigan, 1994.
- Doak, C. M., L. S. Adair, M. Bentley, C. Monteiro, and B. M. Popkin, "The Dual Burden Household and the Nutrition Transition Paradox," *International Journal of Obesity*, 29, 129–36, 2005.

^{© 2019} International Association for Research in Income and Wealth

- Doss, C., "Intrahousehold Bargaining and Resource Allocation in Developing Countries," The World Bank Research Observer, 28, 52–78, 2013.
- Drewnowski, A. and S. E. Specter, "Poverty and Obesity: The Role of Energy Density and Energy Costs," *American Journal of Clinical Nutrition*, 79, 6–16, 2004.

Drèze, J. and A. Sen, Hunger and Public Action, Clarendon, Oxford, 1989.

- Economic Survey of Maharshtra, *Directorate of Economics and Statistics*, Planning Department, Government of Maharashtra, Mumbai, 2018.
- FAO, IFAD, UNICEF, WFP and WHO, *The State of Food Security and Nutrition in the World 2018. Building Climate Resilience for Food Security and Nutrition*, FAO, Rome, 2018. Available at: http://www.fao.org/state-of-food-security-nutrition/en/.
- Fishman, S. M., L. Caulfield, M. deOnis, M. Blossner, A. A. Hyder, L. Mullany, and R. E. Black, "Childhood and Maternal Underweight," in Ezzati, M., A. D. Lopez, A. Rodgers, and C. L. J. Murray (eds), Comparative Quantification of Health Risks: Global and Regional Burden of Disease Attributable to Selected Major Risk Factors, World Health Organization, Geneva, 39– 161, 2004.
- Frankenberger, T., "Indicators and Data Collection Methods for Assessing Household Food Security," in Maxwell, S., and T. Frankenberger (eds), *Household Food Security: Concepts, Indicators, Measurements: A Technical Review*, UNICEF and IFAD, New York and Rome, 77–130, 1992.
- Gopichandran, V., P. Claudius, L. S. Baby, A. Felinda, and V. R. Mohan, "Household Food Security in Urban Tamil Nadu: A Survey in Vellore," *The National Medical Journal of India*, 23, 278, 2010.
- Grantham-McGregor, M., S. Yin Bun Cheung, P. Glewwe, L. Richter, and B. Strupp, "Child Development in Developing Countries 1: Developmental Potential in the First 5 Years for Children in Developing countries (Series)," *The Lancet*, 369, 60–70, 2007
- Greene, W. H., Author's Personal Email Communication with W.H Greene on the Interpretation of Rho in Recursive Bivariate Probit Models, The University of Queensland, Brisbane, Australia, 2010.
 - _____, Econometric Analysis, Prentice Hall PTR, Upper Saddle River, New Jersey, 2012.
- Gupta, P., K. Singh, and V. Sethi, "Food Insecurity Among the Young Children (6–35 months) in Urban Slums of Delhi, India," *Indian Journal of Maternal and Child Health*, 15, 1–6, 2013.
- Gundersen, C., B. Kreider, and J. Pepper, "The Economics of Food Insecurity in the United States," *Applied Economic Perspectives and Policy*, 33, 281–303, 2011.
- Hackett, M., H. Melgar-Quiñonez, and M. C. Álvarez, "Household Food Insecurity Associated with Stunting and Underweight Among Preschool Children in Antioquia, Colombia," *Revista Panamericana de Salud Pública*, 25, 506–10, 2009.
- Haddad, L. The Double Malnutrition Burden: Time for Australia to Lead, Development Policy Blog, 2014. Available at: http://www.devpolicy.org/the-double-malnutrition-burden-time-for-austr alia-to-lead-20140307/.
- Haddad, L., L. Cameron, and I. Barnett, "The Double Burden of Malnutrition in SE Asia and the Pacific: Priorities, Policies and Politics," *Health Policy and Planning*, 30, 1193–206, 2015.
- Haddad, L., J. Hoddinott, and H. Alderman (eds), *Intrahousehold Resource Allocation in Developing Countries: Models, Methods, and Policy*, Johns Hopkins University Press for the International Food Policy Research Institute, Baltimore, MD, 1997.
- Haddad, L., N. Nisbett, I. Barnett, and E. Valli, Maharashtra's Child Stunting Declines: What Is Driving Them? Findings of a Multidisciplinary Analysis, Institute of Development Studies, Brighton, 2014. Available at: https://www.ids.ac.uk/publications/maharashtras-child-stunt ing-declines-what-is-driving-them-findings-of-a-multidisciplinary-analysis/.
- Hamilton, W. L., J. T. Cook, W. W. Thompson, L. F. Buron, E. A. Jr Frongillo, C. M. Olson, and C. A. Wehler, *Household Food Security in the United States in 1995: Technical Report of the Food Security Measurement Project*, US Department of Food and Agriculture, Washington, DC, 1997.
- Hasan, M. M., S. Ahmed, and M. A. H. Chowdhury. "Food Insecurity and Child Undernutrition: Evidence from BDHS 2011," *Journal of Food Security*, 1(2), 52–7, 2013.
- Headey, D. "Turning Economic Growth into Nutrition-Sensitive Growth," Paper presented at the 2020 Conference: Leveraging agriculture for improving nutrition and health, New Delhi, 10–12 February, 2011.
- Headey, D., A. Chiu, and S. Kadiyala, "Agriculture's Role in the Indian Enigma: Help or Hindrance to the Crisis of Undernutrition?," *Food Security*, 4, 7–102, 2012.
- Headey, D. and O. Ecker, "Rethinking the Measurement of Food Security: From First Principles to Best Practice," *Food Security*, 5, 327–43, 2013.

^{© 2019} International Association for Research in Income and Wealth

- Heckman, J. J., "Dummy Endogenous Variables in a Simultaneous Equation System," *Econometrica*, 46, 931–59, 1978.
- Helble, M. and A. Sato (eds), *Wealthy But Unhealthy: Overweight and Obesity in Asia and the Pacific: Trends, Costs, and Policies for Better Health,* Asian Development Bank Institute, Tokyo, 2018.
- Hernandez, D. and A. Jacknowitz, "Transient, But Not Persistent, Adult Food Insecurity Influences Toddler Development1-3," *The Journal of Nutrition*, 139, 1517–24, 2009.
- IIPS-UNICEF, Comprehensive Nutrition Survey of Maharashtra Report, IIPS-UNICEF, Mumbai, 2013. Available at: http://www.iipsindia.org/cnsm_report.htm.
- Isanaka, S., M. Mora-Plazas, S. Lopez-Arana, A. Baylin, and E. Villamor, "Food Insecurity Is Highly Prevalent and Predicts Underweight but Not Overweight in Adults and School Children from Bogotá, Colombia," *Journal of Nutrtion*, 137, 2747–55, 2007.
- Jones, A. D., F. M. Ngure, G. Pelto, and S. L. Young, "What Are We Assessing When We Measure Food Security? A Compendium and Review of Current Metrics," *Advances in Nutrition* (*Bethesda, Md.*), 4, 481–505, 2013.
- Kac, G., M. Schlüssel, R. Pérez-Escamilla, and G. Velásquez-Melendez, "Household Food Insecurity Is Not Associated with BMI for Age or Weight for Height Among Brazilian Children Aged 0-60 Months," *PLoS ONE*, 7, e45747, 2012.
- Levitsky, D. A. and B. J. Strupp, "Malnutrition and the Brain: Changing Concepts, Changing Concerns," *Journal of Nutrition*, 125, S2212–2220, 1995.
- Lewis, M. Investments and Economic Growth: Macroeconomic Evidence and Microeconomic Foundations, 4877, The World Bank, Washington D.C., 2009.
- Maddala, G., *Limited Dependent and Qualitative Variables in Econometrics*, 1st ed, Cambridge University Press, Cambridge, 1983.
- Maitra, C., A Review of Studies Examining the Link Between Food Insecurity and Malnutrition, Technical Paper, FAO, Rome, 2018. Available at: http://www.fao.org/3/CA1447EN/ca1447en.pdf. Masters, G. N., "A Rasch Model for Partial Credit Scoring," Psychometrika, 47, 149–74, 1982.
- Maxwell, S. and M. Smith, "Household Food Security: A Conceptual Review." in Maxwell, S. and T. Frankenberger (eds), *Household Food Security: Concepts, Indicators, Measurements*, IFAD and UNICEF, Rome and New York, 4–77, 1992.
- McDonald, C. M., J. McLean, H. Kroeun, A. Talukder, L. D. Lynd, and T. J. Green, "Household Food Insecurity and Dietary Diversity as Correlates of Maternal and Child Undernutrition in Rural Cambodia," *European Journal of Clinical Nutrition*, 69, 242–6, 2015.
- Meenakshi, J. V., "Trends and Patterns in the Triple Burden of Malnutrition in India," *Agricultural Economics*, 47, 115–34, 2016.
- Menon, P., P. Nguyen, H. Phuong, S. Mani, N. Kohli, R. Avula, and L. M. Tran, *Trends in Nutrition Outcomes, Determinants, and Interventions in India (2006–2016)*, POSHAN Report 10, International Food Policy Research Institute (IFPRI), New Delhi, India, 2017. Available at: http://ebrary.ifpri.org/cdm/ref/collection/p15738coll2/id/131341.
- Mittal, N. and J. V. Meenakshi, "Does the ICDS Improve Children's Diets? Some Evidence from Rural Bihar," *The Journal of Development Studies*, 1–16, 2018.
- Motbainor, A., W. Alemayehu, and K. Abera, "Stunting Is Associated with Food Diversity while Wasting with Food Insecurity among Underfive Children in East and West Gojjam Zones of Amhara Region, Ethiopia," PLoS ONE, 10, e0133542, 2015.
- Mukhopadhyay, D. K. and A. B. Biswas, "Food Security and Anthropometric Failure Among Tribal Children in Bankura, West Bengal," *Indian Pediatrics*, 48, 311, 2011.
- Murray, C. J. and A. D. Lopez, "On the Comparable Quantification of Health Risks: Lessons from the Global Burden of Disease Study," *Epidemiology*, 10, 594–605, 1999.
- National Research Council and Institute of Medicine (NRCIM), Research Opportunities Concerning the Causes and Consequences of Child Food nsecurity and Hunger: A Workshop Summary, N. Kirkendall, C. House, and C.F. Citro, Rapporteurs, Committee on National Statistics, Division of Behavioral and Social Sciences and Education, Food and Nutrition Board, Institute of Medicine, The National Academies Press, Washington DC, 2013.
- National Family Health Survey (NFHS), *The Fourth National Family Health Survey (NFHS-4)*, 2017, International Institute of Population Sciences (IIPS), Mumbai, India, 2017.
- National Sample Survey Organisation (NSSO), *Nutritional Intake in India (Report No. 513, 61ST Round, July 2004—June 2005)*, National Sample Survey Organisation, Ministry of Statistics & Programme Implementation, Government of India, New Delhi, 2007.
- Nettle, D., C. Andrews, and M. Bateson, *Food Insecurity as a Driver of Obesity in Humans: The Insurance Hypothesis*, 40, e105, Cambridge University Press, 2017.
- Ngure, F., M. Reid, M. Brianna, J. H. Humphrey, M. N. Mbuya, G. Pelto, and R. J. Stoltzfus, "Water, Sanitation, and Hygiene (WASH), Environmental Enteropathy, Nutrition, and Early Child Development: Making the Links," *Annals of the New York Academy of Sciences*, 13081, 118–28, 2014.

- Oh, S. Y. and M. J. Hong, "Food Insecurity is Associated with Dietary Intake and Body Size of Korean Children from Low-Income Families in Urban Areas," *European Journal of Clinical Nutrtion*, 57, 1598 – 604, 2003.
- Osei, A., P. Pandey, D. Spiro, J. Nielson, R. Shrestha, Z. Talukder, V. Quinn, and N. Haselow, "Household Food Insecurity and Nutritional Status of Children Aged 6 to 23 Months in Kailali District of Nepal," *Food and Nutrition Bulletin*, 4, 483–94, 2010.
- Osmani, S. R., *The Capability Approach and Human Development: Some Reflections*, UNDP Human Development Report 2016, UNDP, New York, 2016.
- Popkin, B. M., "Global Context of Obesity." in Kumanyika, S., and R. C. Brownson (eds), *Handbook* of Obesity Prevention. Springer, Boston, MA, 2007.
- Quisumbing, A. R., L. R. Brown, H. S. Feldstein, L. Haddad, and C. Peña, "Women: The Key to Food Security," *Food and Nutrition Bulletin*, 17, 1–2, 1996.
- Rah, J. H., A. A. Cronin, B. Badgaiyan, V. M. Aguayo, S. Coates, and S. Ahmed, "Household Sanitation and Personal Hygiene Practices Are Associated with Child Stunting in Rural India: a Cross-Sectional Analysis of Surveys," *British Medical Journal Open*, 5, e005180, 2015.
- Ramalingaswami, V., U. Jonnson, and J. Rohde, "Commentary: The Asian Enigma," in P. Adamson (ed), *The Progress of Nations*, UNICEF, New York, 1996.
- Reis, M., "Food Insecurity and the Relationship between Household Income and Children's Health and Nutrition in Brazil," *Health Economics*, 21, 405–42, 2012.
- Roodman, D., "Estimating Fully Observed Recursive Mixed-Process Models With CMP," Stata Journal, 11, 159–206, 2011.
- Ruel, M. T. and H. Alderman, "Nutrition-Sensitive Interventions and Programmes: How Can They Help to Accelerate Progress in Improving Maternal and Child Nutrition?," *The Lancet*, 382, 536–51, 2013.
- Saha, K. K., E. A. Frongillo, D. S. Alam, S. E. Arifeen, L. A. Persson, and K. M. Rasmussen, "Household Food Security is Associated with Growth of Infants and Young Children in Rural BANGLADESH," *Public Health Nutrition*, 12, 1556–62, 2009.
- Saaka, M. and S. M. Osman, "Does Household Food Insecurity Affect the Nutritional Status of Preschool Children Aged 6–36 Months?," *International Journal of Population Research*, 2013, 1–12, 2013.
- Senauer, B., D. Sahn, and H. Alderman, "The Effect of the Value of Time on Food Consumption Patterns in Developing Countries: Evidence from Sir Lanka," *American Journal of Agricultural Economics*, 68, 920–7, 1986.
- Sen, A. K., "Ingredients of Famine Analysis: Availability and Entitlements," *Quarterly Journal of Economics*, 96, 433–64, 1981.
 - _, Commodities and Capabilities, North-Holland, 1985.
 - , Development as Freedom, Oxford University Press, New York, 1999.
- Sethi, V., Triple Burden of Malnurtion in India: Are We Heading Towards a Human Capital Crisis? Workshop presentation, IIPS-USYD, International Institute for Population Sciences, Mumbai, October 8, 2018.
- Singh, A., A. Singh and R. Faujdar, "Household Food Insecurity and Nutritional Status of Children and Women in Nepal," *Food and Nutrition Bulletin*, 35, 3–11, 2014.
- Skalicky, A., A. Meyers, W. Adams, Z. Yang, J. Cook, and D. A. Frank, "Child Food Insecurity and Iron Deficiency Anaemia in Low-Income Infants and Toddlers in the United States," *Maternal* and Child Health Journal, 10, 177–85, 2006.
- Smith, L. C., "The Great Indian Calorie Debate: Explaining Rising Undernourishment During India's Rapid Economic Growth," *Food Policy*, 50 (C), 53–67, 2015.
- Smith, L., U. Ramakrishnan, A. Ndiaye, L. Haddad, and R. Martorell, "The Importance of Women's Status for Child Nutrition in Developing Countries: International Food Policy Research Institute (IFPRI) Research", *Food and Nutrition Bulletin*, 24, 287–88, 2003.
- Strauss, J., G. Mwabu, and B. Kathleen, "Intrahousehold Allocations: A Review of Theories and Empirical Evidence," *Journal of African Economies*, 9(Supplement1), 83–143, 2000.
- Tarasuk, S. V. and G. H. Beaton, "Women's Dietary Intakes in the Context of Household Food Insecurity," *The Journal of Nutrition*, 129, 672–9, 1999.
- Tarasuk, V., A. Mitchell, L. Andrew, L. McLaren, and L. Mcintyre, "Chronic Physical and Mental Health Conditions Among Adults May Increase Vulnerability to Household Food Insecurity1-3," *The Journal of Nutrition*, 143, 1785–93, 2013.
- Townsend, M. S., J. Peerson, B. Love, C. Achterberg, and S. P. Murphy, "Food Insecurity Is Positively Related to Overweight in Women," *The Journal of Nutrition*, 131, 1738–45, 2001.
- Tripathy, P., N. Nair, S. Barnett, R. Mahapatra, J. Borghi, S. Rath, S. Rath, R. Gope, D. Mahto, R. Sinha, R. Lakshminarayana, V. Patel, C. Pagel, A. Prost, and A. Costello, "Effect of a Participatory Intervention with Women's Groups on Birth Outcomes and Maternal Depression

^{© 2019} International Association for Research in Income and Wealth

in Jharkhand and Orissa, India: a Cluster-Randomised Controlled Trial," *The Lancet*, 375, 1182–92, 2010.

- United Nations Development Programme (UNDP), Human Development Report 1994: New Dimensions of Human Security, UNDP, New York, 1994.
 - , "Africa Human Development Report 2012: Towards a Food Secure Future," *Population and Development Review*, 39, 172–3, 2013.
- UNICEF, Strategy for Improved Nutrition of Children and Women in Developing Countries, Oxford University Press, New York, 1990.
- Velásquez-Melendez, G., M. M. Schlüssel, A. S. Brito, A. A. Silva, J. D. Lopes-Filho, and G. Kac, "Mild But Not Light or Severe Food Insecurity is Associated With Obesity Among Brazilian Women," *Journal of Nutrition*, 141, 898–902, 2011.
- Victora, C. G., L. Adair, C. Fall, P. C. Hallal, R. Martorell, L. Richter, and H. S. Sachdev, for the Maternal and Child Undernutrition Study Group, "Maternal and Child Undernutrition: Consequences for Adult Health and Human Capital," *The Lancet*, 371, 340–57, 2008.
- von Grebmer, K., H. Fritschel, B. Nestorova, T. Olofinbiyi, R. Pandya-Lorch, and Y. Yohannes, *The Global Hunger Index: The Challenge of Hunger 2008*, Bonn, Washington, DC, 2008.
- Wilde, J., "Identification of Multiple Equation Probit Models with Endogenous Dummy Regressors," *Economics Letters*, 69, 309–12, 2000.
- WHO Working Group, "Use and interpretation of anthropometric indicators of nutritional status. WHO Working Group," *Bulletin of World Health Organization*, 64, 929–41, 1986.
- World Food Summit (WFS), Rome Declaration on World Food Security, Rome, 1996.
- World Health Organization (WHO), *Turning the Tide of Malnutrition: Responding to the Challenge of the 21st Century*, World Health Organization, Geneva, 2000. Available at: http://www.who.int/iris/handle/10665/66505.
- _____, Nutrition Landscape Information System (NLIS) Country Profile Indicators: Interpretation Guide, World Health Organization, Geneva, 2010. Available at: http://www.who.int/iris/handl e/10665/44397.
- _____, *The double burden of malnutrition. Policy brief*, World Health Organization, Geneva, 2017. Available at: https://apps.who.int/iris/bitstream/handle/10665/255413/WHO-NMH-NHD-17.3eng.pdf;jsessionxml:id=B7EB64BA2F822003ED708F797CC434F2?sequence=1.

SUPPORTING INFORMATION

Additional supporting information may be found in the online version of this article at the publisher's web site:

 Table A1: Household Food Insecurity Access Scale: Measurement Tool (HFIAS)

Figure B1: Conceptual framework to understand the relationship between maternal/child nutritional status and household food security status.

^{© 2019} International Association for Research in Income and Wealth